### RUTGERS JOURNAL OF LAW & PUBLIC POLICY

VOLUME 6 **FALL 2008** ISSUE 1

Editor-in-Chief

Krishna S. Bhavsar

**Executive Editor** 

DENISE M. MAHER

CHRISTOPHER EMRICH

Managing Editor of Articles

Symposium Editor

RICHARD DAVID WALKER

Managing Technology Editor

DAVID F. ROEBER

**Managing Research** Editor

KATHLEEN M. MCCLURE

**Business Editor** 

SARAH BOYER

Managing Editor of Notes

ELIZABETH POTOSKY

Senior Articles Editor

MATTHEW KOHUT

Lead Technology **Editor** 

RENEE DELLA FAVE

Senior Notes Editor

JOHN C. HYON

**Marketing Editor** 

JULIE PAULL

Submissions Editors

MELISSA ABATEMARCO JESSICA BISIGNANO

Articles Editors

KEVIN GOLDEN **BRANDON HAWKINS** CHARLES HOLMGREN DOUG MAUTE ROBERT RICHARDSON BORIS SHAPIRO JOHN WYAND

**Technology Editors** 

JONATHAN GROPPER BRYCE GUINGRICH

**Notes Editors** 

RICHARD BATELMAN MIKE FORD MARK KOWAL JESS MILLER HEDWIG MURPHY LISA MYERS ADAM ORLACCHIO NOAH SCHWARTZ

Staff Editors

MEGAN HELFRICH ALISON ANDERSON DONALD BENNEDETTO RUSSELL LIEBERMAN GABRIEL BLUESTONE JENNIFER LOWES ROBERT J. CAHALL WARREN MABEY ERIN CARROLL MARIA MANCINI JILLIAN A. CENTANNI PHILIP MERSINGER LEO DONATUCCI PRISCILLA MOORE DAVID MOSCOW ALLISON GROSS AMANDA HARBER NICOLE J. O'HARA MELISSA OSORIO

NICOLAS PEDONE RYAN C. RICHARDS AYA SALEM

SANDY SCHLITT KIMBERLY SCHWARZ NITIN SHARMA JOANNA SYKES-SAAVEDRA

CHRIS WEIR STEVEN WILSON

Faculty Advisors

SARAH E. RICKS

DAMON Y. SMITH

PHILIP L. HARVEY



#### About the Rutgers Journal of Law & Public Policy

The Rutgers Journal of Law and Public Policy (ISSN 1934-3736) is published three times per year by students of the Rutgers School of Law – Camden, located at 217 North Fifth Street, Camden, NJ 08102. The views expressed in the Rutgers Journal of Law & Public Policy are those of the authors and not necessarily of the Rutgers Journal of Law & Public Policy or the Rutgers School of Law – Camden.

**Form**: Citations conform to *The Bluebook: A Uniform System of Citation* (18<sup>th</sup> ed. 2005). Please cite the *Rutgers Journal of Law & Public Policy* as 6 RUTGERS J.L. & PUB. POL'Y \_\_\_(2008).

**Copyright:** All articles copyright © 2008 by the *Rutgers Journal of Law & Public Policy*, except where otherwise expressly indicated. For all articles to which it holds copyright, the *Rutgers Journal of Law & Public Policy* permits copies to be made for classroom use, provided that (1) the author and the *Rutgers Journal of Law & Public Policy* are identified, (2) the proper notice of copyright is affixed to each copy, (3) each copy is distributed at or below cost, and (4) the *Rutgers Journal of Law & Public Policy* is notified of the use.

For reprint permission for purposes other than classroom use, please submit request as specified at http://www.rutgerspolicyjournal.org/.

**Manuscripts**: The *Rutgers Journal of Law & Public Policy* seeks to publish articles making original contributions in the field of public policy. The *Journal* accepts both articles and compelling essays for publication that are related to the expansive topic of public policy. Manuscripts must contain an abstract describing the article or essay which will be edited and used for publication on the website and in CD-ROM format. The *Journal* welcomes submissions from legal scholars, academics, policy makers, practitioners, lawyers, judges and social scientists.

Electronic submissions are encouraged. Submissions by email and attachment should be directed to submissions@rutgerspolicyjournal.org.

Paper or disk submissions should be directed to *Rutgers Journal of Law & Public Policy*, Rutgers University School of Law – Camden, 217 North Fifth Street, Camden, New Jersey 08102.

**Subscriptions**: Subscription requests should be mailed to to *Rutgers Journal of Law & Public Policy*, Rutgers University School of Law – Camden, 217 North Fifth Street, Camden, New Jersey 08102, or emailed to info@rutgerspolicyjournal.org.

**Internet Address**: The *Rutgers Journal of Law & Public Policy* website is located at http://www.rutgerspolicyjournal.org.

#### OFFICERS OF THE UNIVERSITY

RICHARD L. MCCORMICK, B.A., Ph.D., President MARGARET S. MARSH, B.A., M.A., Ph.D., Interim Provost and Dean, FASC

#### LAW SCHOOL ADMINISTRATION

RAYMAN L. SOLOMON, B.A., M.A., J.D., Ph.D., Dean and Professor of Law Jaydev Cholera, B.A., M.B.A., Director, Finance and Administration John S. Beckerman, A.B., M.A., J.D., Ph.D., Associate Dean—Academic Affairs Camille Spinello Andrews, B.A., J.D., Associate Dean—Enrollment Anne V. Dalesandro, A.B., M.L.S., J.D., Director of Law Library Barbara Maylath, A.B., Director of Development Eve Biskind Klothen, B.G.S., J.D., Assistant Dean for Pro Bono and Public Interest Programs Angela V. Baker, B.A., J.D., Associate Dean of Students and Career Planning

#### **FACULTY**

AARON ARI AFILALO, A.B., J.D., LL.M., Associate Professor of Law

DAVID E. BATISTA, A.B., J.D., M.S., Librarian LINDA S. BOSNIAK, A.B., J.D., M.A., Professor of Law

Hon. Dennis Braithwaite, B.S., J.D., LL.M., Visiting Professor of Law

A. HAYS BUTLER, B.A., J.D., M.S. (LIS), Librarian II

LISA CAPASSO, Visiting Clinic Asst. Prof. - LRW MICHAEL A. CARRIER, B.A., J.D., Associate Professor of Law

Fred C. Chandler, Jr., B.A., LL.B., LL.M., Professor of Law Emeritus

GLORIA F. CHAO, B.A., M.L.S., Librarian II EDWARD E. CHASE, B.A., J.D., Professor of Law Emeritus

VICTORIA L. CHASE, B.A., J.D., Clinical Associate Professor of Law

ROGER S. CLARK, B.A., LL.B., LL.M., J.S.D., LL.D., Board of Governors Professor

JASON K. COHEN, B.A., J.D., Legal Writing Clinical Asst. Professor

Russell M. Coombs, B.A., J.D., Associate Professor of Law

LUCY COX, B.A., M.S., Ph.D., M.L.S., Librarian

PERRY DANE, B.A., J.D., Professor of Law JOHN H. DAVIES, B.S., LL.B., LL.M., Professor of Law Emeritus

JAY M. FEINMAN, B.A., J.D., Distinguished Professor of Law

KIMBERLY D. FERZAN, B.S., J.D., Professor of Law

DAVID M. FRANKFORD, B.A., J.D., Professor of Law

Ann E. Freedman, B.A., J.D., Associate Professor of Law

STEVEN F. FRIEDELL, B.A., J.D.,  $Professor\ of\ Law$ 

SANDRA GAVIN, B.A., J.D., Director of Advocacy Programs

ERIC GILSON, B.A., J.D., M.L.I.S., *Librarian III* GEORGE GINSBURGS, Bacchalaureat Serie Mathematiques, B.A., M.A., Ph.D.,

HARRIET N. KATZ, B.A., J.D., Clinical Professor of Law & Director, Lawyering Program

A. KODZO PAAKU KLUDZE, B.A., LL.B., Ph.D.,
Distinguished Professor of Law Emeritus

DONALD KOROBKIN, B.A., A.M., J.D.,  $Professor\ of\ Law$ 

 $\begin{array}{c} {\rm ARTHUR~B.~LABY,\,B.A.,\,J.D.,} \, Associate \\ {\it Professor~of~Law} \end{array}$ 

F. Gregory Lastowka, B.A., J.D., Associate Professor of Law

ARNO LIIVAK, B.A., M.L.S., J.D., Professor of Law Emeritus

HARRY LITMAN, A.B., J.D., Visiting Associate Professor of Law

 $\begin{array}{c} {\rm MICHAEL\,A.\,\,LIVINGSTON,\,A.B.,\,J.D.,\,} Professor\\ of\,Law \end{array}$ 

JOHN C. LORE, III, Visiting Asst. Clinical Professor of Law

JONATHAN MALLAMUD, A.B., J.D., Professor of Law Emeritus

EARL M. MALTZ, B.A., J.D., Distinguished Professor of Law

 $\begin{array}{c} {\rm KIMBERLY\,MUTCHERSON,\,B.A.,\,J.D.,} \ Associate \\ {\it Professor\,of\,Law} \end{array}$ 

JOHN F. K. OBERDIEK, B.A., M.A., J.D., Ph.D., Associate Professor of Law

CRAIG N. OREN, A.B., J.D., Professor of Law TRACI OVERTON, B.A., J.D., Clinical Staff Attorney

DENNIS M. PATTERSON, B.A., M.A., J.D., Ph.D., Distinguished Professor of Law

IMANI PERRY, B.A., J.D., L.L.M., Ph.D., Professor of Law

STANISLAW POMORSKI, LL.B., LL.M., J.S.D.,

Distinguished Professor of Law Emeritus

SARAH PICKS P. A. J.D. Clinical Accordate

SARAH RICKS, B.A., J.D., Clinical Associate Professor, Legal Writing

RUTH ANNE ROBBINS, B.A., J.D., Clinical Professor of Law & Chair, LRW Program SHEILA RODRIGUEZ, Clinical Asst. Professor-LRW

RAND E. ROSENBLATT, B.A., M.Sc., J.D., Professor of Law

PATRICK J. RYAN, B.A., M.A., J.D., LL.M., J.S.D., Associate Professor of Law

- Distinguished Professor of Law Emeritus SALLY F. GOLDFARB, B.A., J.D., Associate Professor of Law
- ELLEN P. GOODMAN, A.B., J.D., Professor of Law
- JOANNE GOTTESMAN, B.A., J.D., Clinical Associate Professor of Law
- PHILLIP L. HARVEY, B.A., J.D., Ph.D., Professor of Law
- ELIZABETH L. HILLMAN, B.S.E.E., J.D., Ph.D.,  $Professorof\ Law$
- N.E.H. HULL, B.A., Ph.D., J.D., Distinguished Professor of Law
- RICHARD HYLAND, A.B., M.F.A., J.D., D.E.A., Distinguished Professor of Law
- JOHN P. JOERGENSEN, B.A., M.A., M.S. (LIS), J.D., Librarian II
- DONALD K. JOSEPH, B.S., LL.B., Visiting Associate Professor of Law

- MEREDITH L. SHALICK, B.A., M.S., J.D., Visiting Clinical Asst. Professor of Law SANDRA SIMKINS, Clinical Associate Professor of Law
- RICHARD G. SINGER, B.A., J.D., LL.M., J.S.D., Distinguished Professor of Law
- DAMON Y. SMITH, B.A., M.U.P., J.D., Assistant Professor of Law
- WILLIAM M. SPEILLER, B.S., LL.B., LL.M., Professor of Law Emeritus
- ALLAN Ř. STEIN, B.A., J.D., Professor of Law BETH STEPHENS, B.A., J.D., Professor of Law Gerardo VII Dostegiji R.A., J.D., Visiting
- GERARDO VILDOSTEGUI, B.A., J.D., Visiting Professor of Law
- CAROL WALLINGER, B.S., J.D., Legal Writing, Clinical Associate Professor of Law
- ROBERT M. WASHBURN, A.B., J.D., LL.M., Professor of Law
- ROBERT F. WILLIAMS, B.A., J.D., LL.M., LL.M., Distinguished Professor of Law

## Rutgers Journal of Law & Public Policy

VOLUME 6 FALL 2008 ISSUE 1

# **Current Issues in Public Policy**



## Rutgers Journal of Law & Public Policy

VOLUME 6 FALL 2008 ISSUE 1

#### **CONTENTS**

CONSTITUTIONAL NIHILISM: POLITICAL SCIENCE AND THE
DECONSTRUCTION OF THE JUDICIARY1
Wayne Batchis
SPORTS AND THE CITY: HOW TO CURB PROFESSIONAL
SPORTS TEAMS' DEMANDS FOR FREE PUBLIC
STADIUMS
Marc Edelman
THE MILITARY COMMISSIONS ACT OF 2006: AN ABJECT
ABDICATION BY CONGRESS
Roger S. Clark
THE SUPREME COURT'S DECISION IN BOUMEDIENE V.
BUSH: THE MILITARY COMMISSIONS ACT OF 2006 AND
HABEAS CORPUS JURISDICTION149
A. Hays Butler
IS TORTURE ALL IN A DAY'S WORK? SCOPE OF
EMPLOYMENT, THE ABSOLUTE IMMUNITY DOCTRINE, AND
HUMAN RIGHTS LITIGATION AGAINST U.S. FEDERAL
OFFICIALS
Elizabeth A. Wilson



# SPORTS AND THE CITY: HOW TO CURB PROFESSIONAL SPORTS TEAMS' DEMANDS FOR FREE PUBLIC STADIUMS

#### Marc Edelman\*

On February 21, 2008, the thirteen commissioners of Miami-Dade County approved a plan to spend \$347 million in taxpayer money to build a new 37,000-seat retractable-roof ballpark for the Florida Marlins baseball club.¹ The plan requires the county to contribute roughly two thirds of the cost for the new ballpark, with the city of Miami contributing three percent (\$10 million)

<sup>\*</sup> Marc Edelman, Esq. (Marc@MarcEdelman.com) is a visiting assistant professor at Rutgers School of Law - Camden and an adjunct professor at New York Law School. Mr. Edelman earned his B.S. in economics from the Wharton School (University of Pennsylvania) and his J.D./M.A. from the University of Michigan. Excerpts from this article have previously appeared in the author's Spring 2003 Virginia Sports & Entertainment Law Review article, *How to Curb Professional Sports' Bargaining Power Vis-à-Vis the American City*, 2 VA. Sports & Ent. L.J. 280 (2003).

<sup>&</sup>lt;sup>1</sup> See Marc Edelman, Marlins Stadium: Deal Expensive, Maybe Unconstitutional, MIAMI HERALD, Mar. 4, 2008, at A19; Sarah Talalay, Play Ball! Marlins Stadium Approved: Commissioners Approve Framework for \$515 Million Deal in Miami, SOUTH FLA. SUN-SENTINEL, Feb. 22, 2008, at 1A ("Miami and Miami-Dade County commissioners approved a binding agreement that serves as the framework for a \$515 million, 37,000-seat, retractable-roof ballpark and a \$94 million parking garage at the site of the Orange Bowl . . . ."). See also Michael Vasquez & Matthew I. Pinzur, Bases Loaded for Marlins: Miami OK's Stadium, MIAMI HERALD, Feb. 21, 2008; Braman v. Miami-Dade County, No. 08-03787-CA-15 1, 5 (Fla. Cir. Ct. Sept. 9, 2008). The stadium building plan was approved as part of a larger Global Interlocal Agreement that also included building a port access tunnel, museum park, streetcar project, and major league soccer stadium. See Braman, No. 08-03787-CA-15 at 3.

and the team owner, Jeffrey Loria, contributing the remaining thirty percent (\$155 million).<sup>2</sup> Loria would then defray most, if not all, of his costs by selling stadium naming rights to a third party.<sup>3</sup>

Many Miami-Dade County residents have objected to this plan.<sup>4</sup> Their biggest complaint is that it allows the Marlins' owner to pay just thirty percent of stadium construction costs, yet entitles him to one hundred percent of stadium revenues.<sup>5</sup> That arrangement hardly seems like an equitable "partnership."

 $<sup>^2</sup>$  Edelman, *Marlins Stadium*, *supra* note 1. *See also* Talalay, *supra* note 1. The Marlins' \$155 million contribution would include \$120 million in upfront spending and another \$35 million in annual rent payments of \$2.3 million a year. *Id*.

<sup>&</sup>lt;sup>3</sup> See Edelman, Marlins Stadium, supra note 1 (noting that providing Loria with the full revenue stream from selling naming rights to the new Marlins ballpark "is essentially free money for Loria, given that the New York Mets recently sold their stadium naming rights to a third party, Citigroup, for \$400 million"). See also Marc Edelman, Why the "Single Entity" Defense Can Never Apply to NFL Clubs: A Primer on Property-Rights Theory in Professional Sports, 18 FORDHAM INTELL. PROP. MEDIA & ENT. L.J. 891, 914 (2008) [hereinafter Edelman, "Single Entity" Defense] (noting the value of stadium naming rights). But see DENNIS R. HOWARD & JOHN L. CROMPTON, FINANCING SPORT 272-86 (2d ed. 2005) (same); Braman, No. 08-03787-CA-15 at 9 (finding the value of naming rights, for some bizarre reason, to be as little as "\$2 million per year").

<sup>&</sup>lt;sup>4</sup> See, e.g., Braman, No. o8-o3787-CA-15. Braman involved a lawsuit seeking to enjoin the public funding of a new Marlins stadium. Id. The court noted "a poll show[ing] that 56% of a segment of the population (super voters) are opposed to the building of the stadium with hotel tax dollars." Id. at 18.

<sup>&</sup>lt;sup>5</sup> See Linda Robertson, Stadium Plan is Bad for City, MIAMI HERALD, July 19, 2008, at D1 ("It's a \$515 million gamble, of which the Marlins are paying one-fifth the cost and reaping 100% of the profits in a sweet deal for Loria, a wealthy art dealer."); Evan S. Benn, Charles Rabin & Michael Vasquez, Miami Megaplan: Megaplan Trial Likely with Talks Going Nowhere - About 70 People Joined Plaintiff Norman Braman on Watson Island to Protest Miami and the County's \$3 Billion Megaplan as Pretrial Settlement Talks Fizzled, MIAMI HERALD, July 13, 2008, at B1 (explaining that under the plan, even "money made from concerts or other sporting events that take place in the park would go directly to the Marlins").

<sup>&</sup>lt;sup>6</sup> See Robertson, supra note 5. The argument that Miami-Dade County's "partnership" with the Marlins is unequal is bolstered by the fact that Marlins ownership has never invested much of its own money into the ballclub. See Larry Larue, Florida Marlins: Good Team, Built Dirt Cheap: Marlins Contend with a Payroll Three-Fourths of A-Rod's Salary, MORNING NEWS TRIBUNE

Nevertheless, many local communities continue to engage in these inequitable "partnerships" because professional sports leagues have monopoly power over the number of franchises in their sport.<sup>7</sup> This monopoly power gives teams the control to switch host communities almost at will, as well as the clout to credibly threaten to switch host communities if a current host fails to meet that team's subsidy demand.<sup>8</sup> Pennsylvania

(Tacoma, Wash.), Aug. 17, 2008, at C3 ("At \$21 million on opening day, their payroll is about \$22 million lower than the 29th team on that list, the Tampa Bay Devil Rays."); Arthur Staple, *Marlins Show They're Not Just Spoilers*, NEWSDAY, May 28, 2008, at A46 (comparing the Marlins' \$21,811,500 opening day payroll for the 2008 season with their rival New York Mets' \$137,793,376 payroll); Dave George, *Trade Could Add Fun to Marlins' Run*, PALM BEACH POST (Fla.), Jul. 31, 2008, at 1C ("Loria certainly isn't shy, not about perturbing fans with a bargain-basement payroll for a constantly rebuilding team . . ."); Carlos Frias, *Marlins Pump Up Volume*, *Not Payroll*, PALM BEACH POST (Fla.), Mar. 31, 2008, at 1A ("[Until the Marlins' new publicly funded ballpark opens], owner Jeffrey Loria has said, there are no plans to start pumping up the lowest payroll in Major League Baseball."); Tom D'Angelo, *Loria: Payroll Increase Will Wait*, PALM BEACH POST (Fla.), Mar. 2, 2008, at 1B ("Jeffrey Loria said Saturday he will boost the salaries, but not until the stadium opens in 2011.").

7 See Michael Leeds & Peter von Allmen, The Economics of Sports 111-12, 154-55 (2d ed. 2005); Rodney Fort, Direct Democracy and the Stadium Mess, in Sports, Jobs & Taxes: The Economic Impact of Sports Teams and STADIUMS 149-50 (Roger G. Noll & Andrew Zimbalist eds., 1997) [hereinafter Fort, Direct Democracy: RODNEY FORT, SPORTS ECONOMICS 140 (2d ed. 2006) [hereinafter FORT, SPORTS ECONOMICS]; Stadium Financing and Franchise Relocation Act of 1999: Hearing on S.952 Before the Sen. Comm. on the Judiciary, 106th Cong. 38 (1999) [hereinafter Hearings] (statement of Andrew Zimbalist) ("An economist has no difficulty in identifying monopoly as the root of the problem."); ANDREW ZIMBALIST, MAY THE BEST TEAM WIN: BASEBALL ECONOMICS AND PUBLIC POLICY 123-24 (2003) [hereinafter ZIMBALIST, MAY THE BEST TEAM WIN]; see generally Marc Edelman & C. Keith Harrison, Analyzing the WNBA's Mandatory Age/Education Policy from a Legal, Cultural, and Ethical Perspective: Women, Men, and the Professional Sports Landscape, 3 NW. J. L. & Soc. Pol'Y 1, 23, 77 (2008) (indicating that even certain other sports leagues such as the WNBA may at times act as monopolists).

<sup>8</sup> See Marc Edelman, How to Curb Professional Sports' Bargaining Power Vis-à-Vis the American City, [hereinafter Edelman, Bargaining Power] 2 VA. SPORTS & ENT. L. J. 280, 284 (2003); Fort, Direct Democracy, supra note 7, at 149-50; see also Braman, No. 08-03787-CA-15 at 6 ("According to [Marlins President David] Samson, the team has publicly stated its intention to relocate if a new baseball stadium is not acquired in the near future."); id. (stating that former President of Major League Baseball Bob DuPuy warned Miami-Dade county commissioners that failure to build the Marlins a new subsidized ballpark would be "a death knell for baseball" in Miami); ZIMBALIST, MAY THE

Senator Arlen Specter describes this practice as "legalized extortion."9

This article argues that America needs to better protect its local communities against monopolist sports leagues' demands for publicly funded stadiums. Part I of this article discusses the evolution of sports stadium subsidies. Part II discusses why American communities continue to provide subsidies to professional sports teams. Part III explains why providing stadium subsidies is a bad idea for most local communities. Part IV discusses four types of proposals intended to reduce sports teams' power to demand stadium subsidies. proposes a federal bill that would better protect the interests of American communities by ensuring that any community which builds a professional sports facility is able to keep the pro rata share of that facility's revenue stream.

#### I. THE HISTORY OF PROFESSIONAL SPORTS **SUBSIDIES**

American communities have not always subsidized the professional sports industry.<sup>10</sup> To the contrary, for the first seventy-five years of professional sports, most team owners built their own facilities and covered their own costs. 11 By the end of

BEST TEAM WIN, supra note 7, at 124 (citing a 2001 letter from Major League Baseball Commissioner, Bud Selig, to former Florida Senator, Alex Villalobos, threatening that, if denied public financing, the Marlins would be relocated).

<sup>9</sup> Hearings, supra note 7, at 11 (testimony of Sen. Arlen Specter of Pennsylvania); see also id. at 53; id. at 31 (testimony of Jean B. Cryor, former Member, Maryland House of Delegates) ("Today team owners are holding the baby captive and waiting for ransom. They are using the fear of losing everything to force the ransom payment.").

<sup>&</sup>lt;sup>10</sup> See Edelman, Bargaining Power, supra note 8, at 284 (describing the "glory era" of professional sports).

<sup>&</sup>lt;sup>11</sup> See id. at 284 (citing Lee Geige, Cheering for the Home Team: An Analysis of Public Funding of Professional Sports Stadia in Cincinnati, Ohio, 30 U. Tol. L. Rev. 459, 461 (1999)); FORT, SPORTS ECONOMICS, supra note 7, at 338. Indeed, until 1950, there were just three publicly funded stadiums used by professional sports teams: the Los Angeles Coliseum (built in 1923), Chicago's Soldier Field (built in 1929) and Cleveland's Municipal Stadium (built in 1931). Edelman, Bargaining Power, supra note 8, at 285; see also John Siegfried &

World War II, however, changing demographics led to the start of communities subsidizing professional sports teams.<sup>12</sup>

#### A. The Emergence of Public Subsidies

The era of publicly funded sports facilities that continues into today began in 1950 when the city of Milwaukee, unable to secure a Major League Baseball ("Baseball") expansion franchise, decided to lure an existing team by building a public stadium.<sup>13</sup>

Enticed by the offer to play in a new, publicly funded stadium, on March 18, 1953 Lou Perini, then the owner of MLB's Boston Braves, decided to move his team to Milwaukee. <sup>14</sup> This move marked the first time since the signing of Baseball's Major League Agreement in 1903 that a MLB team switched host cities. <sup>15</sup>

Andrew Zimbalist, *The Economics of Sports Facilities and Their Communities*, 14 J. Econ. Persp. 95, 95-96 (2000).

- <sup>12</sup> See Edelman, Bargaining Power, supra note 8, at 285 ("With new metropolitan markets in the western United States opened by jet travel, the growth of in-home television, and the baby boomers coming of age, professional sports leagues for the first time encountered significant growth opportunities. Major League Baseball ('MLB'), however, chose not to expand to meet these opportunities.").
- <sup>13</sup> See Siegfried & Zimbalist, supra note 11, at 96; James Quirk & Rodney Fort, Hard Ball: The Abuse of Power in Pro Team Sports 15 (1999) [hereinafter Quirk & Fort, Hard Ball].
- <sup>14</sup> See Siegfried & Zimbalist, supra note 11, at 96; QUIRK & FORT, HARD BALL, supra note 13, at 15; A Fan Site Dedicated to Preserving The Memory of Wisconsin's Lost Treasure, http://www.milwaukeebraves.info (last visited Nov. 12, 2008).
- <sup>15</sup> See James Edward Miller, The Baseball Business: Pursuing Pennants and Profits in Baltimore 31-32 (1990); Jerold J. Duquette, Regulating the National Pastime: Baseball and Antitrust 5, 8 (1999); see generally Marc Edelman, Can Antitrust Law Save the Minnesota Twins? Why Commissioner Selig's Contraction Plan was Never a Sure Deal, 10 Sports L.J. 45, 47 (2003) (discussing the merger of the National and American leagues under the Major League Agreement).

As it turned out, the Braves' move to Milwaukee greatly improved Perini's bottom line. In addition to a new stadium, Perini inherited a larger fan base that purchased 1.8 million tickets in 1953 - more than six times as many tickets as Braves fans bought during the team's final season in Boston. In Boston.

Over the next two years, two other MLB teams, the St. Louis Browns and Philadelphia Athletics, decided to similarly leave shared markets and private stadiums in favor of solo markets and public stadiums.<sup>18</sup> The Browns left St. Louis, a market they shared with the Cardinals, in favor of Baltimore's Memorial Stadium, where they became known as the Baltimore Orioles.<sup>19</sup> Meanwhile, the Athletics, a team which shared the Philadelphia market, moved to Kansas City to play in Municipal Stadium.<sup>20</sup>

In 1958, Brooklyn Dodgers owner Walter O'Malley continued this trend - moving his beloved Dodgers out of Brooklyn and to Los Angeles, another city that had been long trying to land a MLB franchise.<sup>21</sup> Unlike the earlier teams that had moved - the Braves, Athletics, and Browns - the Dodgers had regularly drawn large crowds and maintained a loyal fan base while playing in Brooklyn.<sup>22</sup> However, by moving the Dodgers to Los

<sup>&</sup>lt;sup>16</sup> See Miller, supra note 15, at 31; James Quirk & Rodney D. Fort, Pay Dirt: The Business of Professional Team Sports 480 (1992) [hereinafter Quirk & Fort: Pay Dirt] (Table: Attendance Records: Baseball, National League).

 $<sup>^{17}\,</sup>See$  MILLER, supra note 15, at 31; QUIRK & FORT, PAY DIRT, supra note 16, at 480 (compiling attendance records for Major League Baseball's National League).

 $<sup>^{18}</sup>$  See MILLER, supra note 15, at 79 and accompanying text; QUIRK & FORT, HARD BALL, supra note 13, at 15.

 $<sup>^{19}</sup>$  See Miller, supra note 15, at 79; Quirk & Fort, Hard Ball, supra note 13, at 15.

<sup>&</sup>lt;sup>20</sup> *Id*.

 $<sup>^{21}</sup>$  See Joanna Cagan & Neil DeMause, Field of Schemes: How the Great Stadium Swindle Turns Public Money into Private Profit 186 (1998); Quirk & Fort, Hard Ball, supra note 13, at 16.

<sup>&</sup>lt;sup>22</sup> *Hearings, supra* note 9, at 55 (testimony of Sen. Charles E. Schumer of New York) ("I am one who believes in what Pete Hamill has written[,] that the three most evil men of the 20th century were Hitler, Stalin and Walter O'Malley, Sr.").

Angeles, O'Malley became the beneficiary of a prime chunk of real estate.<sup>23</sup>

## B. The 1960s: A Rollercoaster Ride for Stadium Subsidies

Once O'Malley moved the Dodgers to Los Angeles, MLB owners became cognizant of a basic tenet in economics: the law of supply and demand.<sup>24</sup> As long as there were more cities that wanted to capture the essence of professional sports than there were MLB teams available, existing team owners could levy heavy stadium demands on cities, which would often pay the price.<sup>25</sup> By keeping a limited supply of professional baseball teams, the public share of new stadium financing by the end of the 1950s reached close to 100%.<sup>26</sup>

Shortly thereafter, MLB club owners learned the flip side of this rule, when, in November 1958, New York lawyer William Shea and former Dodgers general manager Branch Rickey announced plans to launch a rival professional baseball league, the Continental League.<sup>27</sup> In fear that the planned rival league would begin to gain a presence for itself in untapped MLB markets, MLB owners quickly announced that they would

<sup>&</sup>lt;sup>23</sup> See Cagan & DeMause, supra note 21, at 186; Quirk & Fort, Hard Ball, supra note 13, at 16. Shortly after O'Malley moved the Dodgers to Los Angles, New York Giants owner Horace Stoneham followed by moving his Giants from Manhattan to San Francisco. The Giants' move, however, was different from the one made by the Dodgers in that the Giants were struggling with attendance before heading to California. See Miller, supra note 15, at 79-80; Quirk & Fort, Pay Dirt, supra note 16, at 480.

 $<sup>^{24}</sup>$   $See\,$  E. Thomas Sullivan & Jeffrey L. Harrison, Understanding Antitrust and its Economic Implications 12-22 (3d ed. 1998) (discussing the law of supply and demand).

 $<sup>^{25}</sup>$  See Fort, Sports Economics, supra note 7, at 140. At the same time, the law of supply and demand has allowed existing team owners to charge huge entry fees to prospective new entrants. See Fort, Direct Democracy, supra note 7, at 7 tbl.1.3 (showing rapidly increasing expansion franchise rights fees).

<sup>&</sup>lt;sup>26</sup> See Siegfried & Zimbalist, supra note 11, at 96; QUIRK & FORT, HARD BALL, supra note 13, at 19-20.

 $<sup>^{27}</sup>$  Duquette, supra note 15, at 52-53;  $see\ also$  Fort, Sports Economics, supra note 7, at 134.

expand into most of the territories that the Continental League sought to enter.<sup>28</sup> From 1962 through 1969, MLB expanded from sixteen to twenty-four teams, temporarily returning the supply of baseball teams back into equilibrium with demand.<sup>29</sup> As a result, the rate of stadium subsidies fell during the early part of the 1960s from nearly 100% to just above 60%.<sup>30</sup>

This counterbalance, however, was short-lived. On August 2, 1960, the Continental League ceased its plans to launch a rival league, and thereafter no other investor group seriously proposed doing the same.<sup>31</sup> As a result, once the next wave of communities interested in investing in professional sports emerged, MLB clubs again had an opportunity to increase their subsidy demands.<sup>32</sup>

#### C. Stadium Subsidies Today

Since the 1970s, most local communities have paid between seventy percent and eighty percent of new stadium building costs.<sup>33</sup> Today, these subsidies are paid to MLB teams, as well as

<sup>&</sup>lt;sup>28</sup> See Duquette, supra note 15, at 53-54; Quirk & Fort, Pay Dirt, supra note 16, at 479-87 (Major League Baseball added eight new teams in the years from 1962-69, with new teams beginning play in New York City, Houston, San Diego, Montreal, Los Angeles, Washington, Seattle and Kansas City); Fort, Sports Economics, supra note 7, at 134, 141 ("[E]xpansion and relocation also protect existing owners from outside competition."); id. at 148 (discussing how leaving viable locations without teams increases the risk of new leagues forming).

<sup>&</sup>lt;sup>29</sup> See generally DUQUETTE, supra note 15.

<sup>&</sup>lt;sup>30</sup> Siegfried & Zimbalist, supra note 11, at 96.

<sup>&</sup>lt;sup>31</sup> See FORT, SPORTS ECONOMICS, supra note 7, at 134; see also Wikipedia, Continental League, http://en.wikipedia.org/wiki/Continental\_League (last visited Nov. 12, 2008).

<sup>&</sup>lt;sup>32</sup> See generally CAGAN & DEMAUSE, supra note 21, at 28-29 ("North America is in the midst of a remarkable stadium and sports arena building boom unlike any other in its history . . . . Between 1980 and 1990, U.S. cities spent some \$750 million on building or renovating sports arenas and stadiums. The bill for the '90s is expected to total anywhere between \$8 billion and \$11 billion, the bulk of it paid by taxpayers – and hidden subsidies could amount to billions more.").

<sup>&</sup>lt;sup>33</sup> See, e.g., Michael Cunningham, Stadium Deal Just Doesn't Make Sense: Money Could Be Used on Much More Important Priorities, SOUTH FLA. SUN-

to teams in the National Football League ("NFL"), National Basketball Association ("NBA") and National Hockey League ("NHL").<sup>34</sup> Although the modern subsidy rate remains below that of the late 1950s, most teams exert more power over local communities today than ever before.<sup>35</sup> This is seen in three ways.

First, local communities today are paying more than ever before to build sports facilities.<sup>36</sup> For instance, the Astrodome, which was the most expensive sports facility built or refurbished prior to 1968, cost \$38,000 (or \$175,000 in 1989 dollars).<sup>37</sup> By contrast, the Skydome, which opened in Toronto, Ontario in 1989, cost \$532 million.<sup>38</sup> In addition, the new publicly funded baseball stadium in Washington, D.C., which opened in April 2008, cost \$611 million,<sup>39</sup> and Indianapolis's new football and NCAA basketball stadium, which opened in August 2008, cost

SENTINEL, Feb. 22, 2008, at 1C (evaluating stadium subsidy percentages since 1992). Cunningham notes that:

There are 20 baseball parks built or that are currently under construction since 1992, when Baltimore's Oriole Park at Camden Yards sparked a building boom. The median public contribution for those parks was 73 percent of costs, according to information compiled by the National Sports Law Institute of Marquette University Law School.

Id.; see also Siegfried & Zimbalist, supra note 11, at 96 tbl.1 (detailing Expenditures on New Sports Facilities for Professional Teams by Decade); FORT, SPORTS ECONOMICS, supra note 7, at 338 (finding, from the period of 2000-06, the median public contribution was just sixty-three percent. Yet, this figure is skewed downward because it includes the San Francisco Giants' entirely privately financed stadium, built in the year 2000).

- 34 See Edelman, Bargaining Power, supra note 8, at 288.
- 35 See infra notes 36-49 and accompanying text.
- <sup>36</sup> See infra notes 37-41 and accompanying text.
- 37 See QUIRK & FORT, PAY DIRT, supra note 16, at 161-63.
- 38 See Quirk & Fort, Pay Dirt, supra note 16, at 161-63.
- <sup>39</sup> See Eric Fisher, In D.C., Baseball Hits a Crossroads at New Park, STREET & SMITH'S SPORTS BUS. J., Mar. 10, 2008, at 18.

\$720 million.<sup>40</sup> According to *Street & Smith's Sports Business Journal*, at the time of publishing this article, the city of Minneapolis is considering building a retractable-roof football stadium that is estimated to cost close to \$1 billion.<sup>41</sup>

In addition, the number of new facilities that sports teams are demanding is also rising.<sup>42</sup> From 1950 to 1959, sports teams moved into only seven new sports facilities, as compared to twenty-one from 1960 to 1969, twenty-five from 1970 to 1979, fourteen from 1980 to 1989, thirty-two from 1990 to 1998, and already more than forty since 1999.<sup>43</sup> A significant percentage of the increase in new stadiums built over the past decade is attributable to baseball and football teams, which once shared community stadiums, beginning to demand their own separate facilities.<sup>44</sup> Another component of the increase is based on teams with greater frequency declaring their facilities obsolete.<sup>45</sup>

<sup>&</sup>lt;sup>40</sup> See Don Muret, Opening in 2008, Street and Smith's Sports Bus. J. (Jan. 21, 2008), at 15; Don Muret, Indy's Showplace is also a Showroom, Street and Smith's Sports Bus. J. (Sept. 15, 2008), at 1 [hereinafter Muret, Indy's Showplace].

<sup>&</sup>lt;sup>41</sup> Don Muret, *Extra 30 Days in Minnesota will Allow for Tour of NFL Stadiums*, STREET AND SMITH'S SPORTS BUS. J. (Sept. 1, 2008), at 24 (estimating the Minnesota stadium potential building cost at \$954 million).

<sup>&</sup>lt;sup>42</sup> See infra note 43 and accompanying text.

<sup>&</sup>lt;sup>43</sup> See Siegfried & Zimbalist, supra note 11, at 96; see also FORT, SPORTS ECONOMICS, supra note 7, at 340 tbl.10.1 (describing Recent and Upcoming Stadium and Arena Openings as of 2004).

<sup>44</sup> See Editorial, Take Us Out, RICHMOND TIMES DISPATCH, Apr. 29, 2008, at A10 ("Orioles Park at Camden Yards started the trend away from multipurpose stadiums shaped like doughnuts and toward baseball-only stadiums with so-called throwback designs."); see also David Armstrong, 49ers on the Move? Economics, Football-Only Stadiums Rarely Pay Off for Cities, Experts Say, SAN FRANCISCO CHRONICLE, Nov. 10, 2006, at B4; Bengals have Sold 20 Percent of Seat Licenses in Three Weeks, Columbus Dispatch (Ohio), Jan. 9, 1997, at 4D (discussing the city of Cincinnati's building of separate ballparks for the Reds and Bengals); Hearings, supra note 9, at 14 (testimony of Sen. Arlen Specter of Pennsylvania) ("[Pennsylvania is] looking at four new [publicly funded] stadiums. Two are under construction now in western Pennsylvania for the Steelers and the Pirates, and two are in the immediate offering for the Phillies and the Eagles.").

<sup>&</sup>lt;sup>45</sup> See, e.g., Siegfried & Zimbalist, supra note 11, at 96; David McLemore, Functioning in a New Arena; San Antonio's Alamodome Alive despite SBC Center, DALLAS MORNING NEWS, Dec. 1, 2002, at 43A.

For example, in 2002, owners of the Spurs basketball team demanded that the city of San Antonio build them a new public arena, even though their current arena was just ten years old.<sup>46</sup>

Finally, in recent years, teams have even begun to negotiate the right to keep sports facilities' non-sports related revenues.<sup>47</sup> For instance, when the State of Maryland in 1998 built its new state-of-the art football stadium for the Baltimore Ravens, Maryland agreed to provide the Ravens' ownership with rights to all of the stadium's revenues, including those derived from rock concerts held during the NFL off-season.<sup>48</sup> Similarly, in Miami-Dade County's recent stadium agreement with Marlins ownership, the county agreed to provide the Marlins owners with 100% of all non-baseball related revenues, including revenues from rock concerts, other sporting events, and even the sale of the stadium's naming rights.<sup>49</sup>

In sum, these three trends have created "such a confusion of interests [that] ordinary tax payers are now expected to subsidize the already immense wealth of . . . an indescribably small number of owners." <sup>50</sup> In other words, subsidized sports stadiums have gone from being an exception in the world of professional sports to something far closer to the rule. <sup>51</sup>

#### D. Today's Sports Teams Do Not Need Subsidies to Turn a Profit

Despite the trend toward subsidizing professional sports stadiums, most professional team owners do not need government aid to profit.<sup>52</sup> This is because, in addition to

<sup>47</sup> See infra notes 48-49 and accompanying text.

<sup>&</sup>lt;sup>46</sup> See McLemore, supra note 45.

<sup>&</sup>lt;sup>48</sup> *Hearings*, *supra* note 9, at 31 (testimony of Jean B. Cryor, former Member, Maryland House of Delegates).

<sup>&</sup>lt;sup>49</sup> See Benn, Rabin & Vasquez, supra note 5.

<sup>&</sup>lt;sup>50</sup> *Hearings*, *supra* note 9, at 13 (testimony of Thomas Finneran, former Speaker of the Massachusetts House of Representatives).

<sup>&</sup>lt;sup>51</sup> See supra notes 11 - 34 and accompanying text.

<sup>&</sup>lt;sup>52</sup> This argument is supported by building a mathematical model for sports-team profitability, which estimates net operating income and annual expected

earning a high rate of return on the team's resale,<sup>53</sup> most team owners have recently learned to capitalize on two important stadium-related revenue streams: stadium naming rights and personal seat licenses.<sup>54</sup>

Stadium naming rights are the rights of corporations to place their name on major sports stadiums.<sup>55</sup> Although the first reputed naming-rights agreement goes back to 1971, when Schaefer Brewing Company paid \$150,000 to name the Patriots' stadium Schaefer Field, sports teams did not begin to recognize the full power of selling naming rights until recently.<sup>56</sup> In recent years, teams in large markets such as the New York Mets have sold stadium naming rights for as much as \$400 million (20 year rights at \$20 million per year).<sup>57</sup> Meanwhile, teams that play in less traditional sports markets such as the Houston Texans have sold their stadium naming rights for as much as \$300 million (30 year rights at \$10 million per year).<sup>58</sup>

return on investment by team. *See generally* Quirk & Fort, Hard Ball, *supra* note 13, at 206, 212. The model considers the additional costs of constructing a new stadium and factors in revenue streams that are created by building a new stadium. *See generally id*.

- 54 See Zimmerman, infra note 79, and accompanying text.
- 55 See HOWARD & CROMPTON, supra note 3, at 272.
- 56 See id.

<sup>&</sup>lt;sup>53</sup> See, e.g., FORT, SPORTS ECONOMICS, supra note 7, at 1-2 (showing that, since 1915, ownership of a sports team such as the New York Yankees has provided twice as large a rate of return as owning a diversified investment portfolio); *id.* at 9 ("Professional team sports generated revenues of about \$13.9 billion in 2002-03.").

<sup>&</sup>lt;sup>57</sup> See Edelman, "Single Entity" Defense, supra note 3, at 914; see also Terry Lefton, CAA Hired to Land Sponsors for the Yankees, Street & Smith's Sports Bus. J., Oct. 1, 2007, at 1; John Lombardo, Barclays-Nets: A Brand Grows in Brooklyn, Street & Smith's Sports Bus. J., Jan. 22, 2007, at 1.

<sup>&</sup>lt;sup>58</sup> See Howard & Crompton, supra note 3, at 275 tbl.7-5 (compiling the largest sports venue naming rights agreements); see also Fort, Sports Economics, supra note 7, at 79 tbl. 3.8 (noting recent naming rights agreements). Among the newest stadium naming rights agreements, the Indiana Stadium and Convention Building Authority recently sold naming rights to their luxurious, new \$720 million facility to Lucas Oil for \$120 million over a 22 year period. See Muret, Indy's Showplace, supra note 40, at 1. The Dallas Mavericks sold their naming rights for \$195 million (30 years at \$6.5 million per year), and the Atlanta Hawks and Thrashers sold naming rights for a

Although many teams that have obtained lucrative naming rights agreements have chosen to build expensive sports facilities, these kind of naming rights agreements could conceivably cover the entire cost of building a more affordable stadium or arena.<sup>59</sup>

Personal seat licenses ("PSLs"), meanwhile, are advance payments to purchase the right to secure a particular seat in a given venue.<sup>60</sup> Although the Dallas Cowboys football team sold a limited number of "seat options" back in 1968,<sup>61</sup> the NFL's Carolina Panthers in 1993 became the first team to extensively use the concept of PSLs when they privately financed their new facility, Bank of America Stadium (formerly known as Ericcson Stadium).<sup>62</sup> By selling PSLs before beginning stadium construction, Carolina Panthers ownership raised \$180 million in upfront capital.<sup>63</sup> Since then, several other sports teams including the Baltimore Ravens, St. Louis Rams, and Chicago

combined \$185 million (20 years at \$9.3 million per year). *See* HOWARD & CROMPTON, *supra* note 3, at 275 tbl.7-5.

<sup>59</sup> For one example of a more affordably built professional sports stadium, Miller Field in Milwaukee, Wisconsin was constructed in time for opening day of the 2001 season at a cost of just \$313 million. See Edelman, Bargaining Power, supra note 8, at 288 (citing Don Walker, Auditors Blame 'Enron-Style Accounting' for Ballpark Cost Dispute; Stadium District Relied on Future Revenue, Memo Says, MILWAUKEE J. SENTINEL, May 22, 2002, at 1A). As another example, the San Francisco Giants privately financed their ballpark, which opened in 2000, for the total cost of \$319 million. Id. at 289 (citing Richard Alm, Nosebleed Prices: Cost of a Day of Baseball Has Soared at New Arenas, Dallas Morning News, Jul. 11, 2000, at 1D). Meanwhile, the Cincinnati Reds new ballpark, which opened in 2003, cost just \$288 million to build. Fort, Sports Economics, supra note 7, at 340.

<sup>60</sup> See Hearings, supra note 9, at 63 (statement of Jerry Richardson, Owner and Founder, Carolina Panthers); see also Hearings, supra note 9, at 104 (testimony of Paul Tagliabue, former Commissioner, National Football League); FORT, SPORTS ECONOMICS, supra note 7, at 42 (discussing the economics behind PSLs).

<sup>&</sup>lt;sup>61</sup> See HOWARD & CROMPTON, supra note 3, at 288.

<sup>&</sup>lt;sup>62</sup> *Hearings*, *supra* note 9, at 63 (statement of Jerry Richardson, Owner and Founder, Carolina Panthers); *see also* HOWARD & CROMPTON, *supra* note 3, at 288.

<sup>&</sup>lt;sup>63</sup> See Hearings, supra note 9, at 63 (statement of Jerry Richardson, Owner and Founder, Carolina Panthers).

Bears have copied this strategy, similarly raising substantial amounts of money.64

#### II. WHY AMERICAN COMMUNITIES SUBSIDIZE PROFESSIONAL SPORTS

Although professional sports teams rarely need subsidies to profit, most American communities continue to subsidize their professional sports teams because they fear their teams would otherwise move to other communities. 65 As explained by sports economist Rodney Fort, "[l]eaving some viable locations without teams enhances the bargaining power of existing owners with their current host cities."66 Some examples of viable locations without teams in each of the four premier sports leagues include: Los Angeles (no NFL team), Houston (no NHL team), Seattle (no NBA or NHL team), Cleveland (no NHL team), San Diego (no NBA or NHL team), St. Louis (no NBA team), Pittsburgh (no NBA team) and Baltimore (no NBA or NHL team).67

In a perfectly competitive market, new premier leagues such as Shea and Rickey's Continental League would periodically emerge to meet communities' demand for sports teams, and existing leagues would in turn have an incentive to expand into on-hold cities.<sup>68</sup> However, in practice, the four premier sports

<sup>64</sup> See HOWARD & CROMPTON, supra note 3, at 288 tbl. 7-7 (compiling statistics on the size, price, and economic magnitude of current PSL programs).

<sup>65</sup> See Fort, Direct Democracy, supra note 7, at 149-50; FORT, SPORTS ECONOMICS, supra note 7, at 140 ("Owners acting through their leagues limit the number of teams in their league by choice rather than through the forces of competition. There are two important indicators that the number of teams is smaller than a more economically competitive sports world would give to fans. First, rival leagues do form occasionally . . . . Second, every time a league announces that it plans to expand, a long line of candidate-owners forms in hope of becoming the newest addition to the league.").

<sup>&</sup>lt;sup>66</sup> FORT, SPORTS ECONOMICS, supra note 7, at 141.

<sup>&</sup>lt;sup>67</sup> See Leeds & von Allmen, supra note 7, at 100 tbl.3.7 (listing the fifteen most populous metropolitan areas and their respective sports teams).

<sup>68</sup> See generally Duquette, supra note 15, at 52-53; Fort, Sports ECONOMICS, supra note 7, at 140.

leagues rarely face competition from any new league because sports markets have high barriers to entry.<sup>69</sup> Indeed, competitor leagues are rarely able to compete against MLB or the NBA, NFL and NHL because these leagues enjoy an almost insurmountable lead in building a fan base, signing superstar players, acquiring television broadcast contracts,<sup>70</sup> and obtaining playing facilities.<sup>71</sup> For this reason, some liken a sports league's tight control on its number of franchises to a form of blackmail or extortion.<sup>72</sup>

According to former Washington, D.C. mayor Sharon Pratt Kelly, the limited number of franchises in professional sports forces American communities to deal with "a prisoner's dilemma of sorts."<sup>73</sup> The dilemma is that "if no mayor succumbs to the demands of a franchise shopping for a new home, then the team will stay where they are." <sup>74</sup> However, this outcome is unlikely because "if Mayor A is not willing to pay the price,

<sup>69</sup> See Edelman, Bargaining Power, supra note 8, at 291. Almost all attempts to form rival, premier professional sports leagues in the four major sports over the past forty years have failed. For example, in football, the World Football League emerged in the 1960s and almost from its first games exhibited dire financial trouble; missed payrolls were common and the league folded during its second season. See ROBERT BERRY ET AL., LABOR RELATIONS IN PROFESSIONAL SPORTS 93 (1986). The United States Football League then emerged in 1983 with a differentiated strategy of playing football during the spring; the USFL quickly found itself in a bidding war for players with the NFL, and in November of 1984, it too filed for bankruptcy. See id. at 95-96. In basketball, Harlem Globetrotters owner Abe Saperstein launched the American Basketball League in 1961, but the league folded in its second season. See id. at 155. Its successor, the American Basketball Association, which started in 1967-68, performed slightly better; however, it too was heading toward bankruptcy in 1976 when the league disbanded and four existing teams joined the NBA. Id. at In hockey, the World Hockey Association was founded by entrepreneurs Gary Davidson and Dennis Murphy in 1972, but by 1979, most of its teams were bankrupt; the four remaining franchises were acquired by the NHL. *Id.* at 213-14.

<sup>&</sup>lt;sup>70</sup> See Leeds & von Allmen, supra note 7, at 124-25.

<sup>&</sup>lt;sup>71</sup> See Quirk & Fort, Hard Ball, supra note 13, at 135.

<sup>&</sup>lt;sup>72</sup> See, e.g., id. at 6; Hearings, supra note 9, at 11 (testimony of Sen. Arlen Specter of Pennsylvania).

<sup>&</sup>lt;sup>73</sup> Fort, *Direct Democracy*, supra note 7, at 150.

<sup>74</sup> *Id*.

Mayor B may think it is advantageous to open up the city's wallet. Then to protect his or her interest, Mayor A often ends up paying the demanded price."75

#### III. WHY STADIUM SUBSIDIES ARE A BAD IDEA FOR LOCAL COMMUNITIES

Absent the monopoly power of America's professional sports leagues, few communities would likely subsidize professional sports industry.<sup>76</sup> Although rooting professional sports teams is often a source of personal enjoyment, stadium funding rarely adds economic value to a local community.<sup>77</sup> In addition, the social benefits of new public stadiums are often misaligned in favor of the already wealthy.<sup>78</sup>

#### A. Sports Stadiums Add No Economic Value to The **Local Community**

It is often claimed by both sports owners and local government officials that public stadiums add great financial value to local communities.<sup>79</sup> For instance, Carolina Panthers

75 *Id*.

<sup>&</sup>lt;sup>76</sup> As empirical support for this statement, one needs to look no further than professional soccer in Europe, where the Associations Europeennes de Football imposes no restrictions on league entry, and as a result, teams regularly privately finance their own stadiums, or else relinquish significant rights over team assets to the public bodies that assist them. See STEFAN SZYMANSKI & ANDREW ZIMBALIST, NATIONAL PASTIME: HOW AMERICANS PLAY BASEBALL AND THE REST OF THE WORLD PLAYS SOCCER 130-31 (2005); see also Michael Smith, IMG Venture to Offer Private Financing for College Facilities, STREET & SMITH'S SPORTS BUS. J., Aug. 25, 2008, at 3 ("ISG, a London-based joint venture between IMG and Bastion Stadiums LLP, has used the private-financing model to pay for construction of Wembley Stadium [in England] as well as projects in Brazil and India.").

<sup>77</sup> See infra notes 79-94 and accompanying text.

<sup>&</sup>lt;sup>78</sup> See infra notes 101-11 and accompanying text.

<sup>&</sup>lt;sup>79</sup> See Dennis Zimmerman, Subsidizing Stadiums: Who Benefits, Who Pays?, in Sports, Jobs & Taxes: The Economic Impact of Sports Teams and STADIUMS 120-25 (1997); see also John L. Crompton, Economic Impact Analysis of Sports Facilities and Events: Eleven Sources of Misapplication, 9 J. SPORT MGMT, 14, 15 (1995); CLEAN v. State, 928 P.2d 1054, 1057 (Wash. 1996)

owner Jerry Richardson once stated that "[c]ities use stadiums to facilitate and accomplish a wide variety of purposes [including serving] as the centerpiece of a broader urban renovation plan, and . . . enhancing local communities and their economies." Similarly, former Houston Mayor Lee Brown has claimed that building a football stadium and bringing professional football to Houston would "generate millions of dollars for our city." Meanwhile, according to the Miami commissioners that approved publicly subsidizing a new Marlins ballpark, the new ballpark will soon help to economically revitalize Little Havana, an area surrounded by working class and low income housing. 82

Independent, empirical research, however, consistently has rebutted these claims, finding no positive correlation between facility construction and economic development.<sup>83</sup> One prominent research study, conducted by the Congressional Research Service, analyzed 30 different stadium projects and found that none of them positively impacted the local community.<sup>84</sup> Another study, conducted by economists Robert Baade and Allen Sanderson, found that subsidized sports facilities also do not improve local employment.<sup>85</sup> Meanwhile, a

("Robbie Stern, Special Assistant to the President of the Washington State Labor Council, expanded on the view of the business persons, stressing the broader impact of the Mariners on the state's economy, saying, 'Here is an opportunity to use tax money to create family wage jobs . . . . ").

<sup>&</sup>lt;sup>80</sup> *Hearings*, *supra* note 9, at 61 (testimony of Jerry Richardson, Owner and Founder, Carolina Panthers).

<sup>&</sup>lt;sup>81</sup> FORT, SPORTS ECONOMICS, *supra* note 7, at 338, 339.

<sup>&</sup>lt;sup>82</sup> Braman v. Miami-Dade County, No. 08-03787-CA-15 1, 12-13 (Fla. Cir. Ct. Sept. 9, 2008).

<sup>83</sup> See generally Siegfried & Zimbalist, supra note 11, at 103.

<sup>&</sup>lt;sup>84</sup> See generally Todd Senkiewicz, Stadium and Arena Financing: Who Should Pay?, 8 SETON HALL J. SPORT L. 575, 589 (1998) (citing William J. Donovan, Stadiums: Winners or Losers, PROVIDENCE JOURNAL-BULLETIN, Feb. 26, 1997, at A1).

<sup>&</sup>lt;sup>85</sup> See Robert A. Baade & Allen R. Sanderson, The Employment Effect of Teams and Sports Facilities, in Sports, Jobs, and Taxes: The Economic Impact of Sports Teams and Stadiums 93 (Roger G. Noll & Andrew Zimbalist eds., 1997) (describing how spending on spectator sports "is largely offset by reductions in other forms of leisure spending by consumers and other fiscal

third study, prepared by the Maryland Department of Business and Economic Development,<sup>86</sup> found that the Baltimore Ravens' new stadium cost more money and created fewer jobs than the best alternative public tax investment, and the stadium did not even lead to much in-state tourist spending.<sup>87</sup>

A simple visit to some of the communities that have recently invested in new sports stadiums seems to buttress these conclusions. For example, according to Jean Cryor, a former member of the Maryland House of Delegates, although the State of Maryland was promised by Ravens officials that building a new stadium would provide Baltimore with "a wave of tourism money," in reality, most fans that travel to Baltimore to watch a Ravens game then drive directly home afterwards. Similarly, according to Dr. Phillip Porter, an expert in sports economics from the University of South Florida, money spent at the ballpark generally does not spread to local businesses because "stadiums... are self-contained." In reaching this conclusion, Dr. Porter relied on a study he undertook of Tropicana Field, which was built in Tampa for the Rays baseball team.

commitments by government entities"); see also Mark S. Rosentraub, Stadiums and Urban Space, in Sports, Jobs, and Taxes: The Economic Impact of Sports Teams and Stadiums 186 (Roger G. Noll & Andrew Zimbalist eds., 1997) (stating that even some paid consulting studies concede that the substitution effect does exist). Although there are studies available by paid consulting firms that claim new ballparks create thousands of jobs annually, those studies, which are self-serving, generally involve "an inappropriate methodology, based on antiquated input-output models and unrealistic assumptions." ZIMBALIST, MAY THE BEST TEAM WIN, supra note 7, at 125.

<sup>&</sup>lt;sup>86</sup> The Maryland Department of Business and Economic Development is part of the state's executive branch. This study was afterward reviewed by the Maryland Department of Fiscal Services – part of the state's legislative branch. *See* Zimmerman, *supra* note 79, at 122-23.

<sup>87</sup> See Zimmerman, supra note 79, at 122-23.

<sup>88</sup> See infra notes 89-92 and accompanying text.

 $<sup>^{89}</sup>$   $Hearings,\ supra$  note 9, at 31 (testimony of Jean B. Cryor, former Member, Maryland House of Delegates).

<sup>&</sup>lt;sup>90</sup> Braman v. Miami-Dade County, No. 08-03787-CA-15 1, 13 (Fla. Cir. Ct. Sept. 9, 2008).

<sup>91</sup> *Id*.

According to Porter, even fifteen years after the stadium was constructed, there is still little neighborhood development, and the surrounding area remains mostly abandoned.92

In summary, as testified before Congress by sports economist Andrew Zimbalist, who now consults for MLB,93 "most of the money that gets spent [on sports facilities] is [simply] re-circulated money within the town. It does not generate new value added."94

#### B. Social Benefits of Stadium Subsidies are Misaligned

Other supporters of stadium subsidies concede that building sports facilities do not in themselves revitalize the economy; yet, they still support stadium subsidies as way to provide social value.95 Some argue that professional sports teams provide communities with a public good, the ability to "root for the home team," and with a level of community spirit that would not exist in the absence of sports.96 As explained by Seattle radio

<sup>92</sup> Id. at 13; see also id. at 15 ("Dr. Porter has studied 13 stadiums nationwide and concluded that only four garnered a positive impact for the local community, while the remaining nine actually had a negative impact according to the sales tax data.").

<sup>93</sup> See Nathan Vardi, A Royal Mess (Professional Baseball, Kansas City Royals), FORBES, May 7, 2007, at 40; Jeff Nash, Turnaround Job: Making Tigers' Overnight Success Last, CRAIN'S, DET. BUS. Oct. 16, 2006, at 1; Jeff Passan, Slice of Strife - Royals say they need their \$55 million to compete, KANSAS CITY STAR, Jan. 27, 2006, at D1.

<sup>94</sup> Hearings, supra note 9, at 37 (testimony of Andrew Zimbalist).

<sup>95</sup> See Martin Greenberg, Sports Facilities and Development: Stadium Financing and Franchise Relocation Act of 1999, 10 MARQ. SPORTS L.J. 383, 387 (2000); CLEAN v. State, 928 P.2d 1054, 1067 (Wash. 1996) (accepting the state's argument that loss of the Seattle Mariners baseball club from Seattle would "diminish[] the quality of life for a substantial number of [the] state's citizens); Hearings, supra note 9, at 66 (testimony of Benjamin Klein, Professor of Economics, University of California, Los Angeles) ("One must take into account what economist call the public good consumption benefits of these projects."); Braman, No. 08-03787-CA-15 at 17 (restating the contention of Miami-Dade County Mayor Carlos Alvarez that: "For children, for tourists, professional sports brings people of all walks of life together").

<sup>96</sup> See Zimmerman, supra note 79, at 121.

talk show host Vincent Richini, the Mariners playing in Seattle provides "something that's intangible . . . it's a way of life."97

Another form of this same argument claims that professional sports teams help communities to develop national identities, increasing the morale of communities that host sports teams. Mayors of what have traditionally been defined as second-tier cities such as Charlotte, Nashville and Oklahoma City have defended paying subsidies as a way to solve their cities' long-term identity crises. For example, on the day after Oklahoma City residents voted in favor of creating a 15-month, one-cent sales tax to fund renovating the city's basketball arena and lure away the Seattle Supersonics, Oklahoma City Mayor Mick Cornett declared "the arena vote would help put Oklahoma City on a world stage." 100

The problem with both of these policy arguments, however, is that the social benefits of building public stadiums skew in favor of the wealthy, whereas the costs skew in favor of the middle- and lower-class. <sup>101</sup> In other words, government spending on sports facilities fails to follow the benefits principle of taxation - the principle that each taxpayer's contribution to provide for a public service should remain in proportion to the benefits received from that service. <sup>102</sup>

<sup>97</sup> CLEAN, 928 P.2d at 1054.

<sup>&</sup>lt;sup>98</sup> See Senkiewicz, supra note 84, at 593 (citing Mark S. Rosentraub, Major League Losers: The Real Cost of Sports and Who's Paying for It 19 (1997)).

<sup>&</sup>lt;sup>99</sup> See id. at 593 (referencing Charlotte and Nashville as second-tier cities); see also Oklahoma City Passes Arena Tax to Lure Sonics, SEATTLE TIMES, Mar. 5, 2008, at A1 (referencing Oklahoma City).

<sup>&</sup>lt;sup>100</sup> Oklahoma City Passes Arena Tax to Lure Sonics, supra note 99, at A1.

<sup>&</sup>lt;sup>101</sup> See Zimmerman, supra note 79, at 122.

<sup>&</sup>lt;sup>102</sup> See Zimmerman, supra note 79, at 120-21 (stating that "to implement the benefit principle of taxation, revenue must be raised or spending on other programs decreased in such a manner that the distribution of stadium costs matches the distribution of stadium benefits").

#### 1. Social Benefits of Stadium Subsidies

The main social benefit of stadium subsidies - the opportunity to attend games in the new stadium - skews in favor the wealthy because the current cost of attending a professional sports events precludes the lower- and middle-income segment of the population.<sup>103</sup> For example, in 2008, the average price for a family of four to attend a New England Patriots football game was nearly \$600.<sup>104</sup> Meanwhile, the average price for the same family to attend a Chicago Bears game was close to \$500.<sup>105</sup> For the common American, this means that taking his family to either one of these events would cost roughly one percent of that family's pre-tax income.<sup>106</sup>

Recent stadium-design trends have only further priced the typical American out of the professional sports game market. For example, new seating innovations such as luxury boxes and premium seating have led to building new stadiums that contain a greater number of seats targeted for the ultra-wealthy. In turn, many new stadiums have reduced their number of traditional seats.

#### 2. Costs of Stadium Subsidies

By contrast, the tax burden of building new sports facilities often falls predominantly on the lower and middle classes.<sup>108</sup> From 1990 until 2003, sales taxes, which are regressive in

<sup>103</sup> *Id*. at 121.

<sup>&</sup>lt;sup>104</sup> See Scott Van Voorhis & Greg Turner, Pats Passes Priciest: Team Tops NFL with \$118 Average Ticket Price, BOSTON HERALD, Sept. 6, 2008, at 18 (the exact projected price is \$596; this price includes four median price tickets (\$471); two small beers (\$15); four small soft drinks (\$16); four hot dogs (\$14); two game programs (\$10); two caps (\$30) and parking (\$40)).

<sup>&</sup>lt;sup>105</sup> *Id*.

<sup>&</sup>lt;sup>106</sup> See Roger Lehecka & Andrew Delblanco, *Ivy-League Letdown*, N.Y. TIMES, Jan. 22, 2008, at A21 (noting "the median family income in the United States is around \$50,000").

 $<sup>^{107}</sup>$  See Zimmerman, supra note 79, at 121; ZIMBALIST, MAY THE BEST TEAM WIN, supra note 7, at 130.

<sup>&</sup>lt;sup>108</sup> See generally Zimmerman, supra note 79, at 121.

nature, accounted for 29 percent of the public funds going to new stadium construction.<sup>109</sup> Meanwhile, in the most recent stadium building cycle, sales tax increases were used to fund new stadiums in Arlington, Cincinnati, Denver, Indianapolis, Oklahoma City and Tampa.<sup>110</sup> In addition, local lotteries, which are also regressive in impact, were used in Seattle and in the State of Maryland to subsidize their new sports facilities.<sup>111</sup>

Some of the more recent stadium construction plans, such as the one recently approved by Miami-Dade County, have attempted to address this concern by raising funds for new sports stadiums exclusively from tourist taxes. However, even where stadium funding comes entirely from tourist taxes, there is still an opportunity cost in the loss of the opportunity to raise funding for an alternative project that may help the community at large, rather than a project that primarily advantages already wealthy team owners and upper-class fans.

## IV. RECENT PROPOSALS TO CURB STADIUM SUBSIDIES

Over the past several years, politicians and academics have proposed many solutions to help curb sports teams' excessive bargaining power in demanding stadium subsidies.<sup>112</sup> These proposals generally fall into four broad categories: enforcing state lending of credit and public purpose doctrines; ordering league expansion; requiring breakup of the big leagues (divestitures); and implementing congressional statutes.<sup>113</sup>

TANGERS SO

-

<sup>&</sup>lt;sup>109</sup> ZIMBALIST, MAY THE BEST TEAM WIN, *supra* note 7, at 130.

<sup>&</sup>lt;sup>110</sup> See Senkiewicz, supra note 84, at 585-86; see also 8 Projects to Watch in 2008, Street & Smith's Sports Bus. J., Jan. 16, 2008, at 16 (explaining Indianapolis's use of various sales taxes including a food and beverage tax and a restaurant tax); Oklahoma City Passes Arena Tax to Lure Sonics, supra note 99, at A1 (referencing the use of a sales tax in Oklahoma City).

<sup>&</sup>lt;sup>111</sup> See Zimmerman, supra note 79, at 126 (citing Charles T. Clotfelter & Phillip J. Cook, Selling Hope: State Lotteries in America (1989)); see also Senkiewicz, supra note 84, at 586.

<sup>&</sup>lt;sup>112</sup> See infra notes 114-227 and accompanying text.

<sup>&</sup>lt;sup>113</sup> *Id*.

## A. Enforcing State Lending-of-Credit and Public Purpose Doctrines

State lending-of-credit and public purpose doctrines are the oldest and generally most conservative way to address the problem of business demands for public spending.<sup>114</sup> These doctrines arose alongside the growth of big business in the mid-1800s, as citizens became enraged at state and local governments for subsidizing powerful business interests such as the railroad industry.<sup>115</sup>

Since the mid-1800s, forty-six states have applied these doctrines to prohibit the use of public money to aid private enterprise. Courts, however, have rarely applied these doctrines against the building of public sports facilities. 117

#### 1. Lending-of-Credit Doctrines

Lending-of-credit doctrines arise from state enactments that prohibit state and local governments from lending credit to private enterprises. For example, Article III, Section 34 of the Maryland Constitution states, in pertinent part, that "[t]he credit of the State shall not in any manner be given, or loaned to, or in aid of any individual association or corporation." Similarly, Article VIII, Sections 5 and 7 of the Washington Constitution provide that the state and its political subdivisions may not give or loan credit to any private individual or

<sup>&</sup>lt;sup>114</sup> See Edelman, Bargaining Power, supra note 8, at 297. Generally, advocates of state autonomy support curbing professional sports' bargaining power via state statutes because it allows each state to independently determine whether to subsidize sports facilities. *Id.* at n.151.

<sup>115</sup> Id.

<sup>&</sup>lt;sup>116</sup> Dale F. Rubin, Public Aid to Professional Sports Teams—A Constitutional Disgrace: The Battle to Revive Judicial Rulings and State Constitutional Enactments Prohibiting Public Subsidies to Private Corporations, 30 U. Tol. L. Rev. 393, 412 (1999).

 $<sup>^{117}\</sup> See\ infra$  notes 122-27 and accompanying text, and notes 132-48 and accompanying text.

<sup>&</sup>lt;sup>118</sup> See generally Rubin, supra note 116, at 397-99.

<sup>119</sup> MD. CONST. art. III, § 34.

corporation.<sup>120</sup> Meanwhile, Article VII, Section 10 of the Florida Constitution states that "[n]either the state nor any county... shall become a joint owner with, or stockholder of, or give, lend or use its taxing power or credit to aid any corporation, association, partnership or person..."<sup>121</sup>

Although the language of lending-of-credit doctrines seems on its face to prohibit publicly subsidizing professional sports facilities, <sup>122</sup> courts have repeatedly found ways to avoid applying these doctrines in that context. <sup>123</sup> For example, the Washington Supreme Court, in the case *CLEAN v. State*, <sup>124</sup> found that Article VIII, Sections 5 and 7 of the Washington Constitution actually permitted the public funding of a new stadium for the Seattle Mariners because the stadium was to be owned and managed by public government, and the Mariners were required to pay "reasonable rent." <sup>125</sup> The court in *CLEAN* did not even inquire into how to determine if a particular rent is "reasonable." <sup>126</sup> It simply deferred to the judgment of those same executives who had decided in the first instance to build a new ballpark. <sup>127</sup>

#### 2. Public Purpose Doctrines

Public purpose doctrines, meanwhile, arise from state enactments that require governments to use public money only for public purposes.<sup>128</sup> For example, Article VII, Section 1 of the

 $<sup>^{120}</sup>$  Wash. Const. art. VIII, §§ 5, 7. See also CLEAN v. State, 928 P.2d 1054, 1061 (Wash. 1996).

<sup>121</sup> FLA. CONST. art. VII, § 10.

<sup>&</sup>lt;sup>122</sup> See Rubin, supra note 116, at 413.

 $<sup>^{123}</sup>$  See infra notes 124-27 and accompanying text; see also CLEAN, 928 P.2d at 1062; Libertarian Party of Wis. v. State, 546 N.W.2d 424 (Wis. 1996); Rubin, supra note 116, at 412-16.

<sup>124</sup> CLEAN, 928 P.2d at 1061.

<sup>125</sup> *Id.* at 1062.

<sup>126</sup> See id.

<sup>&</sup>lt;sup>127</sup> *Id.* ("In our judgment, a plain reading of the Stadium Act reveals no intent by the Legislature to donate public funds to the Seattle Mariners.").

<sup>&</sup>lt;sup>128</sup> Rubin, *supra* note 116, at 417.

Washington Constitution states that taxes "shall be levied and collected for public purposes only." Similarly, Article VII, Section 9 of the Florida Constitution prevents counties from levying taxes for anything other than "municipal purposes." 130

Much like with enforcing lending-of-credit doctrines, state courts have been hesitant to apply public purpose doctrines against stadium subsidies.<sup>131</sup> For instance, in *CLEAN*, the Supreme Court of Washington rejected the argument that spending tax dollars on the public development of a baseball stadium as a home field for the Seattle Mariners violates Article VII, Section 1 of the Washington Constitution. 132 Washington Supreme Court instead found "construction of a publicly owned stadium to be leased to professional sports teams serves a public purpose" as long as it "confer[s] a benefit of reasonably general character to a significant part of the public."133 While the court noted that it is "not unmindful of the fact that the Seattle Mariners may also reap benefits as the principal tenant of the publicly owned stadium . . . . [t]he fact that private ends are incidentally advanced is immaterial to determining whether legislation furthers a public purpose."134

<sup>129</sup> WASH. CONST. art. VII, § 1.

<sup>130</sup> FLA. CONST. art. VII, § 9.

<sup>&</sup>lt;sup>131</sup> See infra notes 132-53 and accompanying text.

<sup>&</sup>lt;sup>132</sup> CLEAN v. State, 928 P.2d 1054, 1059, 1061 (Wash. 1996).

<sup>133</sup> Id. at 1059-60.

<sup>&</sup>lt;sup>134</sup> *Id.* at 1061 (citing United States v. Town of N. Bonneville, 621 P.2d 127 (1980)). The court further went on to state — in direct opposition to almost all independent academic analyses of the issue — that "public provision of a venue for professional sports franchises serves a public purpose in that the presence in a community of a professional sports franchise provides jobs, recreation for citizens, and promotes economic development and tourism." *Id.* Although the plaintiffs submitted several articles explaining that the economic benefits to a community of building a sports stadium are negligible, the court found that determining whether sports stadiums provide economic benefits is an issue that "is better left to the Legislature, which can assess the relative merits of the arguments for and against a proposition." *Id.* at 1066. Based on the discussion in Part III of this paper, it is difficult to fathom how the court could have earnestly reached that conclusion. *See supra* notes 76-111 and accompanying text.

Until recently, it seemed that, unlike Washington, the State of Florida was one state that still was willing to apply its "public purpose" doctrine against funding the construction of sports stadiums. Specifically, in the 1966 case, *Brandes v. City of Deerfield Beach*, <sup>135</sup> the Florida Supreme Court found that the City of Deerfield Beach's proposal to build a spring training facility for the Pittsburgh Pirates baseball team violated Florida's public purpose doctrine. <sup>136</sup> According to the court in *Brandes*,

[t]he mere incidental advantage to the public resulting from a public aid in the promotion of [a professional sports team] is not a public or municipal purpose; and the incidental benefits or advantages gained by private enterprise from expenditures made for a public purpose do not vitiate or diminish the public purpose.<sup>137</sup>

Nevertheless, two of the more recent Florida "public purpose" cases, *Poe v. Hillsborough County*, <sup>138</sup> and *Braman v. Miami-Dade County*, <sup>139</sup> have all but stripped away the holding in *Brandes*. <sup>140</sup> First, in *Poe*, the Florida Supreme Court upheld public funding of a football stadium for shared use by the Tampa Bay Buccaneers and various public high school teams. <sup>141</sup> At the time, it seemed the decision in *Poe* may have implicitly differentiated *Brandes* on two factual grounds: (1) the publicly financed stadium in *Poe* was to be used for certain public events, including high school and college football games, and (2) the stadium built in *Poe* was intended to attract additional revenue

<sup>135</sup> Brandes v. City of Deerfield Beach, 186 So. 2d 6 (Fla. 1966).

<sup>136</sup> Id. at 7, 12.

<sup>137</sup> Id. at 12.

<sup>&</sup>lt;sup>138</sup> Poe v. Hillsborough County, 695 So. 2d 672 (Fla. 1997).

<sup>&</sup>lt;sup>139</sup> Braman v. Miami-Dade County, No. 08-03787-CA-15 1, 5 (Fla. Cir. Ct. Sept. 9, 2008).

<sup>&</sup>lt;sup>140</sup> See infra notes 141-151 and accompanying text.

<sup>141</sup> Poe, 695 So. 2d at 679.

by hosting the Super Bowl, a major tourist attraction. The Florida Supreme Court, however, never explained these differentiations in its opinion.

Then, in *Braman* (decided on Sept. 9, 2008), a Florida lower court upheld Miami-Dade County's decision to build a baseball stadium for the Florida Marlins, even though the stadium agreement only entitled Miami-Dade County to use of the stadium for sixteen days per year. The *Braman* court reached this ruling based on a broad application of *Poe* and, essentially, the rejection of *Brandes*. 144

According to *Braman*, under Florida law "the burden is on the Plaintiff to prove that [a stadium] project . . . fails to serve a paramount public purpose." <sup>145</sup> Based on this standard, a court "must give deference to the Commission as long as it has competent substantial evidence for its decision . . . ." <sup>146</sup> The *Braman* court ultimately found "ample legislative findings in the evidence . . . for a finding that . . . the building of a [Marlins] stadium constitutes a paramount public purpose." <sup>147</sup> Nevertheless, the court conceded that "the Marlins are getting what amounts to a sweet deal." <sup>148</sup>

<sup>&</sup>lt;sup>142</sup> See id. at 678.

<sup>&</sup>lt;sup>143</sup> According to the lease, the Marlins have full stadium rights for 349 days per year, while "[t]he County and City will have the right to use the stadium for 16 days each year for community events." *Braman*, No. 08-03787-CA-15 at 9. The Marlins have first choice on selecting days. *See id*.

<sup>&</sup>lt;sup>144</sup> See id. at 15. The Florida lower court refers to the facts surrounding the Marlins subsidy, and perhaps all other modern subsidies for Florida sports teams, as "strikingly similar" to *Poe*, and different from *Brandes*. *Id.* at 27, 34. The *Braman* court first distinguished *Brandes* because that case was decided in 1966, "when there were no legislative declarations by the state that the construction of a sports facility served a public purpose." *Id.* at 34. In addition, the court found it relevant that "the sports facility in *Brandes* served an out-of-state sports team providing them with a spring training facility for exhibition games." *Id.* 

<sup>145</sup> *Id*. at 20.

<sup>&</sup>lt;sup>146</sup> *Id*. at 23.

<sup>147</sup> *Id.* at 34.

<sup>148</sup> Id. (emphasis added).

As a practical matter, the *Braman* holding helps to show just how unlikely it is today that any court would strictly interpret its state's "public purpose" doctrine against the building of a sports stadium. Indeed, the opinion in *Braman* spends more than a full page citing to cases in which courts from other states have upheld building sports stadiums under their state's public purpose doctrine, a matter seemingly of great implicit importance. According to the court in *Braman*, "[w]hile the holdings of courts from other jurisdictions are not determinative of the issue . . . these cases are certainly instructive . . . in deciding [the] issue . . . ."151

Based on the Florida state court's ruling in *Braman*, as well as similar rulings reached by many other state courts, it seems futile to continue attempting to regulate professional sports subsidies through state law, as this approach only shifts the "prisoner's dilemma" discussed above by former Washington, D.C. mayor Sharon Kelly from state executives to state judges. <sup>152</sup> A federal-backed solution, applying to all fifty states, therefore seems like a far better approach. <sup>153</sup>

## **B.** Court-ordered Expansion of Professional Sports Leagues

A second potential way to curb professional sports teams' bargaining power involves using federal antitrust law to order the four premier sports leagues to expand their total number of

<sup>&</sup>lt;sup>149</sup> See generally QUIRK & FORT, HARD BALL, supra note 13 (discussing the prisoner's dilemma).

<sup>&</sup>lt;sup>150</sup> Braman v. Miami-Dade County, No. 08-03787-CA-15 1, 32-34 (Fla. Cir. Ct. Sept. 9, 2008).

<sup>&</sup>lt;sup>151</sup> *Id.* at 35-36.

<sup>&</sup>lt;sup>152</sup> See Edelman, Bargaining Power, supra note 8, at 299 ("This argument fails to consider that part of the reason why states are so unwilling to strictly interpret their constitutions is because individual states fear that in doing so they would lose professional sports teams to more leniently-interpreting states.").

<sup>153</sup> See id.

teams.<sup>154</sup> This approach would lead to an increased supply of sports teams per league, thus better balancing team supply with community demand.<sup>155</sup> As a result, it would remove from team owners the ability to credibly threaten to relocate to a different market if their host community denies subsidies.<sup>156</sup>

Thus far, only two plaintiffs have attempted to obtain court-ordered entry into an existing professional sports league. Both plaintiffs, however, have failed on their claims.<sup>157</sup>

#### 1. Mid-South Grizzlies v. National Football League

In the first case, *Mid-South Grizzles v. National Football League*, the plaintiff, Mid-South Grizzlies Limited Partnership, brought an antitrust suit in the Eastern District of Pennsylvania against each of the individual NFL teams, arguing that the teams' refusal to accept the Grizzlies' membership into the league illegally restrained trade. The reviewing court in that case ultimately ruled in favor of the NFL teams, holding that "a professional sport league's refusal to accept for membership a qualified applicant for a franchise in an area where no current league team is located" does not violate either Section 1 or Section 2 of the Sherman Act. 159

More specifically in *Mid-South Grizzlies*, the Grizzlies ownership group filed three separate antitrust claims. The first claim contended that the NFL teams' collective decision to deny them entry into the NFL amounted to an illegal group boycott

63

<sup>&</sup>lt;sup>154</sup> Section 1 of the Sherman Act reads in part: "Every contract, combination in the form of trust or otherwise, or conspiracy, in restraint of trade or commerce among the several States . . . is declared to be illegal." 15 U.S.C. § 1 (2004). Section 2 of the Sherman Act provides: "Every person who shall monopolize, or attempt to monopolize, or combine or conspire with any other person or persons, to monopolize any part of the trade or commerce among the several States . . . shall be deemed guilty of a felony . . . ." 15 U.S.C. § 2 (2004).

<sup>&</sup>lt;sup>155</sup> See generally FORT, SPORTS ECONOMICS, supra note 7, at 141.

<sup>156</sup> *Id*.

<sup>&</sup>lt;sup>157</sup> See infra notes 158-78 and accompanying text.

<sup>&</sup>lt;sup>158</sup> Mid-South Grizzlies v. NFL, 550 F. Supp. 558, 560-61, 562 (E.D. Pa. 1982), *aff'd* 720 F.2d 772 (3d Cir. 1983).

<sup>159</sup> *Id.* at 560.

under Section 1 of the Sherman Act.<sup>160</sup> With respect to that claim, the district court ruled in favor of the NFL based on the Rule of Reason.<sup>161</sup> The court explained that the NFL's decision to limit its number of teams gave the Grizzlies and other rejected franchise applicants the "motivation to form a rival league" that could compete against the NFL teams.<sup>162</sup> Thus, the court found the restraint pro-competitive.<sup>163</sup> The Third Circuit thereafter affirmed.<sup>164</sup>

The Grizzlies' second claim contended that the NFL clubs again violated Section 1 of the Sherman Act by conspiring to deny the Grizzlies access to an essential facility in the form of league membership. Here, the district court again found in favor of the NFL clubs, concluding that the Grizzlies' "reliance on [the essential facilities doctrine was] misplaced" because the essential facilities doctrine "is applicable only where a party is being denied access to something necessary for that party to

<sup>&</sup>lt;sup>160</sup> See id. at 565-66.

<sup>&</sup>lt;sup>161</sup> *Id.* In assessing whether the allegedly anticompetitive conduct violates Section 1 of the Sherman Act, a court will begin with a *prima facie* review of the conduct by applying one of the Supreme Court's three sanctioned tests. Edelman & Harrison, *supra* note 7, at 38. If the allegedly anticompetitive conduct seems to yield an ambiguous effect, a court will apply Rule of Reason analysis, under which a court conducts a full economic investigation. *Id.* at 39. Applying the Rule of Reason, a court will determine whether a plaintiff can make a *prima facie* showing of a violation based on the presence of the following three factors: (1) market power, (2) anticompetitive effects that exceed any procompetitive justifications, and (3) harm. *Id.* 

<sup>&</sup>lt;sup>162</sup> Mid-South Grizzlies, 550 F. Supp. at 568.

<sup>&</sup>lt;sup>163</sup> *Id*.

<sup>&</sup>lt;sup>164</sup> See Mid-South Grizzlies, 720 F.2d 772, 786-87 (noting, however, that the court of appeals did not accept the NFL's position that there is no intra-league competition).

<sup>&</sup>lt;sup>165</sup> See Mid-South Grizzlies, 550 F. Supp. at 569. The "essential facilities doctrine" is applicable where a party is denied access to something necessary to engage in business which is controlled by his competitors. *Id.* at 569-70. Under this doctrine, courts have required certain joint ventures to adopt objective membership rules, giving all parties equal opportunity to participate in the venture. Edelman, *Bargaining Power*, *supra* note 8, at 303. Courts have held essential facilities to include railroad hubs, bridges and ferries, access to news stories, local telephone lines, and, even in certain circumstances, stadium usage. *Id.* at 303-04 (citing several cases).

engage in business which is controlled by his competitors." <sup>166</sup> According to the court, the Mid-South Grizzlies were not competitors to the other NFL clubs in the market for football clubs because "plaintiffs wish to join with defendants, not to compete with them." <sup>167</sup> On this issue, the Third Circuit Court of Appeals again affirmed. <sup>168</sup>

Finally, the Grizzlies' third antitrust claim contended that the NFL clubs unlawfully monopolized the professional football market in violation of Section 2 of the Sherman Act.<sup>169</sup> Here, the district court similarly found no violation because it found no abuse by the NFL. The district court concluded that "[p]laintiffs simply are not rivals or potential rivals of defendants except on the playing field," and that the Grizzlies are "still free to promote a rival league," as well as that the NFL clubs' actions "have done nothing to prevent the formation of a rival league or the fielding of a team in Memphis, Tennessee." Thereafter, the Third Circuit Court of Appeals affirmed, noting that "the Memphis home team market has been left by the NFL for potential competitors. Thus on this record summary judgment on the section 2 Sherman Act claim was also proper." The state of the NFL for potential competitions.

# 2. Seattle Totems Hockey Club, Inc. v. National Hockey League

In the second case, *Seattle Totems Hockey Club, Inc. v. National Hockey League*, a Seattle hockey club, which had been a member of the defunct World Hockey League and was thereafter denied entry into the NHL, brought a Section 2 claim in the Ninth Circuit against the NHL teams.<sup>172</sup> Much like the Third Circuit's findings in *Mid-South Grizzlies*, the Ninth Circuit

<sup>&</sup>lt;sup>166</sup> Mid-South Grizzlies, 550 F. Supp. at 569 (emphasis removed).

<sup>&</sup>lt;sup>167</sup> *Id.* at 570.

<sup>&</sup>lt;sup>168</sup> *Mid-South Grizzlies*, 720 F.2d at 786-87.

<sup>&</sup>lt;sup>169</sup> See Mid-South Grizzlies, 550 F. Supp. at 571.

<sup>170</sup> *Id*.

<sup>171</sup> Mid-South Grizzlies, 720 F.2d at 788.

<sup>&</sup>lt;sup>172</sup> Seattle Totems Hockey Club, Inc. v. NHL, 783 F.2d 1347 (9th Cir. 1986).

rejected the Totems' monopolization claim, finding that "any denial of an NHL franchise to plaintiffs . . . was procompetitive, rather than anti-competitive in its effect and hence not violative of the anti-trust laws." <sup>173</sup>

According to the Ninth Circuit, for "the plaintiff [to succeed on its Section 2 claim, it] must prove that the defendant's actions caused a decrease in competition in the relevant market." However, according to the court, the Totems "were not competing with the NHL; they were seeking to join it." In addition, the Ninth Circuit found no proof that the Totems were denied membership to protect another major league team because, at the time, the Seattle market had no NHL hockey team.

Based on this holding, as well as the similar holding in *Mid-South Grizzlies*, it seems clear that a plaintiff's best chance, if any, to obtain court-ordered expansion into an existing sports league would involve seeking a court order to enter a market that already contains a team in that league.<sup>177</sup> Nonetheless, when reading *Mid-South Grizzlies* and *Seattle Totems* alongside other antitrust cases involving the Rule of Reason, the chances are remote that even under the same-market scenario any court would order a sports league to expand.<sup>178</sup>

# C. Applying Antitrust Law to Break up Professional Sports Leagues

A third potential way to curb professional sports clubs' bargaining power would be to impose a court-ordered breakup

<sup>176</sup> *Id.* Indeed, this leaves open the possibility that the Ninth Circuit may reach a different ruling if a team seeks to enter a sports league in a market where one of the league's teams already exists. *See id.* 

<sup>&</sup>lt;sup>173</sup> *Id.* at 1350 (quoting unpublished district court opinion).

 $<sup>^{174}</sup>$  Id. (quoting Aydin Corp. v. Loral Corp., 718 F.2d 897, 902 (9th Cir. 1983)) (footnote omitted).

<sup>&</sup>lt;sup>175</sup> *Id*.

<sup>&</sup>lt;sup>177</sup> See supra notes 169-70 and accompanying text.

<sup>&</sup>lt;sup>178</sup> See supra notes 158-76 and accompanying text.

of the four premier sports leagues.<sup>179</sup> By breaking up the big leagues, economic incentives for teams and leagues would change so that sports leagues would have the enticement to expand as quickly as possible to maximize league revenues, rather than keep potential host cities on hold without a team.<sup>180</sup>

Any proposal to break up the four premier sports leagues would require that courts apply similar logic to the professional sports market as was applied to the telecommunication market in the famous breakup of AT&T's long-term monopoly over telephone services.<sup>181</sup> It would first require that the Department of Justice bring a monopolization charge against the four premier sports leagues.<sup>182</sup> Then, a federal court would need to hold that the leagues violated antitrust laws, and order the violation remedied by separating MLB, the NBA, NFL and NHL into smaller, competitor leagues.<sup>183</sup>

The argument to break up the professional sports leagues is a favorite amongst academics in both the areas of law and economics. In the famous 1997 article, *Economic Impact of Sports Teams and Facilities*, two distinguished economics

<sup>&</sup>lt;sup>179</sup> See Rodney Fort, Employment Effect of Teams and Sports Facilities, in Sports, Jobs, and Taxes: The Economic Impact of Sports Teams and Stadiums 88 (Roger G. Noll & Andrew Zimbalist eds., 1997); Fort, Sports Economics, *supra* note 7, at 445-49; Quirk & Fort, Hard Ball, *supra* note 13, at 177; Paul C. Weiler & Gary R. Roberts, Sports and the Law: Text, Cases, Problems 194-95 (2d ed. 1998).

<sup>&</sup>lt;sup>180</sup> See FORT, SPORTS ECONOMICS, supra note 7, at 140.

 $<sup>^{181}</sup>$  See Weiler & Roberts, supra note 179, at 643 (citing United States v. Am. Tel. and Tel. Co., 552 F. Supp. 131 (D.D.C. 1982)).

<sup>&</sup>lt;sup>182</sup> See generally QUIRK & FORT, HARD BALL, supra note 13, at 177.

<sup>&</sup>lt;sup>183</sup> See Stephen D. Ross, Monopoly Sports Leagues, 73 MINN. L. REV 643 (1989); Weiler & Roberts, supra note 179, at 643. Usually, a movement to break up the four premier sports leagues would require courts to consider sports leagues under Section 2 of the Sherman Act; if the court considers this path, there are two kinds of charges available under Section 2 of the Sherman Act. Id. at 596. A monopolization charge consists of possession of monopoly power and use of unacceptable means to acquire or maintain that power. Id. An "attempt to monopolize" claim requires a showing that the defendant is dangerously close to obtaining market power and is taking steps with the specific attempt to obtain a monopoly. Id.

<sup>&</sup>lt;sup>184</sup> See infra notes 185-90 and accompanying text.

professors, Roger Noll and Andrew Zimbalist, concluded that if there were multiple leagues per sport, professional sports' excessive bargaining power would dissipate. According to Noll and Zimbalist, in a multiple league model, each league would expand into as many markets as could support a team, and each league would place more teams into the markets that could support multiple teams. As Zimbalist later testified to Congress, "[i]f you had the AL and NL competing in business terms in baseball [as entirely separate league structures], you wouldn't have had Washington DC, one of the 10 largest media markets in the United States, without a baseball team." 187

Concurring with Noll and Zimbalist's position are economics professors James Quirk and Rodney Fort, who in their 1999 book *Hard Ball* conclude that there are substantial market benefits to creating competing leagues in each premier, professional sport.<sup>188</sup> In addition, sports law professors Paul Weiler and Gary Roberts, co-authors of the field's leading textbook *Sports and the Law*, also recognize that the economics of the multiple-league model would remove incentives for teams to demand government subsidies.<sup>189</sup> Weiler and Roberts explain that the competitive pressures generated by multiple sports leagues would further lead to innovations that would enhance fans' overall enjoyment of sports, such as those that occurred

<sup>&</sup>lt;sup>185</sup> Roger Noll & Andrew Zimbalist, *Economic Impact of Sports Teams and Facilities*, in Sports, Jobs and Taxes: The Economic Impact of Sports Teams and Stadiums 65, 88 (1997).

<sup>186</sup> NOLL & ZIMBALIST, *supra* note 185, at 88. Since accepting a consulting position with MLB, Zimbalist has seemed to soften on his stance against public funding for sports stadiums. *See* Andrew Zimablist, *Rays' Owners Offer Worthy Finance Plan for St. Petersburg*, Street & Smith's Sports Bus. J., Dec. 17, 2007, at 46 (supporting a new stadium-building proposal where the Tampa Bay Rays organization contributes "roughly \$150 million, or one-third of the construction costs of the new ballpark"). However, Zimablist's seemingly softened position goes against the weight of his research conducted throughout his career, as well as positions that he has previously taken while testifying under oath. *See*, *e.g.*, *Hearings*, *supra* note 9, at 38-40 (testimony of Andrew Zimbalist).

<sup>&</sup>lt;sup>187</sup> Hearings, supra note 9, at 38 (testimony of Andrew Zimbalist).

<sup>188</sup> See QUIRK & FORT, HARD BALL, supra note 13, at 177.

<sup>&</sup>lt;sup>189</sup> See id.; see also Weiler & Roberts, supra note 179, at 594.

when the American Basketball Association briefly competed against the NBA and introduced the slam dunk and three-point shot.<sup>190</sup>

Nevertheless, there are also real problems with breaking up professional sports leagues. The main problem with breaking up the big four professional leagues is that historically the free market has failed to sustain multiple, premier leagues competing against one another in the same sport. Although the current sample size is too small to conclude with certainty that a competitive sports league model would never work, any long-term attempts in America at multiple-league competition, thus far, have either led to a merger of the leagues or else failure of at least one of the leagues. Such a result may be inevitable. Historically, in a multi-league model, one league generally gains a comparative advantage and drives the others out of business—even if all of the leagues begin operating at the same time and with similar resources. Alternatively, both leagues may

<sup>190</sup> WEILER & ROBERTS, supra note 179, at 595.

<sup>&</sup>lt;sup>191</sup> See, e.g., Mid-South Grizzlies v. NFL, 550 F. Supp. 558, 563 (E.D. Pa. 1982), aff'd 720 F.2d 772 (3d Cir. 1983), ("The NFL has its roots early in this century. Throughout the years several other leagues appeared usually only for a short time without much success."); DAVID QUENTIN VOIGHT, THE LEAGUE THAT FAILED (The Scarecrow Press, Inc. 1998) (discussing the Inter-League Wars in professional baseball during the early 1890s, which eventually led to the survival of just one professional baseball league); Jody Goldstein, Alive & Kicking in the U.S.A., HOUS. CHRON., June 5, 1994, at 21 (discussing how competition between the National Indoor Soccer League and the Major Indoor Soccer League helped drive most teams in both leagues into bankruptcy); Lena Williams, Former Team Official Recounts the A.B.L.'s Dizzying Descent, N.Y. TIMES, Apr. 2, 1999, at D5 (discussing how the rivalry between America's two premier women's basketball leagues resulted in the Women's National Basketball Association driving the American Basketball League into bankruptcy).

<sup>&</sup>lt;sup>192</sup> See Fort, Sports Economics, supra note 7, at 151 (Table 5.4 Pro Sports League Rivalries) (showing the ultimate fate of rival professional sports leagues in the premier professional sports); but see Joe Nocera, It's a Bloody Takeover, N.Y. Times, Sept. 12, 2008, at MM-34 (mentioning that the Premier League and the Football League have both survived in England, even after the 20-team Premier League broke away from the 92-team Football League).

<sup>&</sup>lt;sup>193</sup> See Edelman, Bargaining Power, supra note 8, at 302; see also FORT, SPORTS ECONOMICS, supra note 7, at 153-56. Indeed, there are a few exceptions; for example, in the early 1960s, the American Football League seemed to survive as a rival to the NFL, until 1966 when the leagues merged into one. See

simultaneously drive each other out of business by both overpaying for players and advertisements in attempts to become the dominant league.<sup>194</sup>

This "survival of the fittest" dynamic in professional sports appeared subsequent to brief eras of competition in 1890s baseball, 1980s men's soccer, and late 1990s women's basketball. 195 A likely reason for it is that premier sports leagues are different from other forms of business in that they require the services of elite members of their labor force in order to maintain "major league" status. According to Bill Veeck, arguably the greatest owner and promoter in professional sports history, operating a professional sports team involves marketing dreams —intangible benefits identified with the best players, coaches and managers. 196 These dreams are only obtainable through extraordinarily high labor-market competition. 197

Additional reasons why breaking up the big leagues may fail include the logistical costs involved in the breakup, <sup>198</sup> as well as the high start-up costs of constructing new sports facilities absent strong guarantees of league sustainability, and the fact that sports fans seem to enjoy the tradition that underlies the way leagues currently operate. <sup>199</sup> Unlike consumers of

also Mid-South Grizzlies, 550 F. Supp. at 563; FORT, SPORTS ECONOMICS, supra note 7, at 151. In addition, baseball's National League and American League effectively coexisted until they silently merged together into MLB in 1903. See id

<sup>&</sup>lt;sup>194</sup> See Edelman, Bargaining Power, supra note 8, at 302; see also FORT, SPORTS ECONOMICS, supra note 7, at 156.

<sup>&</sup>lt;sup>195</sup> See supra note 191 and accompanying text.

<sup>&</sup>lt;sup>196</sup> See generally QUIRK & FORT, HARD BALL, supra note 13, at 4.

<sup>&</sup>lt;sup>197</sup> See generally id. at 135. Indeed, there is probably a sustainable market to create rival sports on the minor league level. See Edelman, Bargaining Power, supra note 8, at 302 (citing ARTHUR T. JOHNSON, MINOR LEAGUE BASEBALL AND LOCAL ECONOMIC DEVELOPMENT 1-35 (Univ. of Ill. Press 1995). However, in many ways, this market already appears to be highly saturated. See Edelman, Bargaining Power, supra note 8, at 302.

<sup>198</sup> See generally Edelman, Bargaining Power, supra note 8, at 303.

<sup>&</sup>lt;sup>199</sup> See, e.g., Play at the Plate: Fans Want 'Devil Ray' to Stay, TAMPA BAY BUS. J. (May 1, 2007), available at 2007 WLNR 8219824 (tradition sensitive fans oppose team's proposed name change); Ron Chimelis, Baseball Displays Smarts, THE REPUBLICAN, Nov. 17, 2005, at C1 ("Baseball fans treat the game's

telecommunications services, most consumers of professional sports place significant value on maintaining historical consistency and tradition within the game — something they might not want to sacrifice, even if it were to lead to a personal cost savings.<sup>200</sup>

### **D. Congressional Statutes**

Finally, a fourth potential way to reduce professional sports leagues' bargaining power is through congressional statutes.<sup>201</sup> Over the past fifteen years, two proposals that were intended to address sports leagues' excessive bargaining power were New York Senator Daniel Patrick Moynihan's 1996 Stop Tax-Exempt Arena Debt Issuance Act ("STADIA"),<sup>202</sup> and Pennsylvania Senator Arlen Specter's 1999 Stadium Financing and Franchise Relocation Act ("SFFRA").<sup>203</sup> Neither of these two proposals, however, ever went to a congressional vote.<sup>204</sup> In addition, even if passed, neither of these proposals would have fully resolved the problem of sports stadium subsidies.<sup>205</sup>

The first of these proposals, STADIA, was intended to prevent tax-exempt federal bonds from financing recreational facilities managed by private authorities.<sup>206</sup> Ultimately, Congress rejected STADIA for various reasons, including that

traditions and records as sacred . . . . "); Thomas J. Fitzgerald & Colleen Mancino, *N.J. Has the Other Sports, So . . . 'Why not Baseball?' Local Fans Speak Out: Bring the Yankees West, N.J. Rec., Aug. 9, 1999, at A3 ("Still, baseball fans worship tradition, and some said safety and convenience couldn't outweigh [that].").* 

<sup>&</sup>lt;sup>200</sup> Fitzgerald & Mancino, supra note 199, at A3.

<sup>&</sup>lt;sup>201</sup> See infra notes 202-18 and accompanying text.

<sup>&</sup>lt;sup>202</sup> See Andrew H. Goodman, The Public Financing of Professional Sports Stadiums: Policy and Practice, 9 Sports Law. J. 173, 217-19 (2002); see also Weiler & Roberts, supra note 179, at 265.

<sup>&</sup>lt;sup>203</sup> S. 952, 106th Cong. (1999).

<sup>&</sup>lt;sup>204</sup> Weiler & Roberts, supra note 179, at 630-31.

<sup>&</sup>lt;sup>205</sup> See infra notes 208, 216-18 and accompanying text.

 $<sup>^{206}</sup>$  See Goodman, supra note 202, at 217-219; see also Weiler & Roberts, supra note 179, at 265.

the bill had an adverse effect on zoos and libraries.<sup>207</sup> However, even if Congress had enacted STADIA, the bill would have failed to prevent American cities from financing professional sports stadiums via tax-exempt local bonds.<sup>208</sup> As a result, STADIA would have done little to cure the actual problem of teams demanding subsidies, as long as other government financing arrangements still remained available.

SFFRA, meanwhile, intended to use antitrust principles, including threats of repealing the Sports Broadcasting Act of 1961,<sup>209</sup> to limit the amount of money that local governments could contribute to financing professional baseball and football facilities.<sup>210</sup> Specifically, SFFRA would have required both MLB and NFL teams, in order to maintain their antitrust exemption for pooling sports broadcasts,<sup>211</sup> to place ten percent of their national broadcasting revenues into a trust fund reserved for stadium financing, and contribute 50 percent of the cost of new stadiums from that fund, with individual team owners paying at

The antitrust laws... shall not apply to any joint agreement by or among persons engaging in or conducting the organized professional team sports... by which any league of clubs sells or otherwise transfers all or any part of the rights of such league's member clubs in the sponsored telecasting of the games... engaged in or conducted by such clubs.

Sports Broadcasting Act, 15 U.S.C. § 1291 (2000); see also WEILER & ROBERTS, supra note 179, at 641; FORT, SPORTS ECONOMICS, supra note 7, at 384 ("Under the Sports Broadcasting Act of 1961, league-wide TV contracts... are exempted from antitrust. This exemption continued to the present day, despite the fact that the Supreme Court struck down precisely the same type of behavior by the NCAA."); Edelman, Single Entity, supra note 3, at 918 n.162.

<sup>&</sup>lt;sup>207</sup> Goodman, *supra* note 202, at 218; *see also* Weiler & Roberts, *supra* note 179, at 630-61 (noting that Senator Moynihan's retirement and death may have played a role in the abrupt loss of focus on ending tax-exempt bond spending on sports facilities).

<sup>&</sup>lt;sup>208</sup> See Goodman, supra note 202, at 217-18.

<sup>&</sup>lt;sup>209</sup> The Sports Broadcasting Act of 1961 allows professional sports teams in the four premier professional sports leagues to pool their broadcast rights without risk of violating Section 1 of the Sherman Act.

<sup>&</sup>lt;sup>210</sup> See Hearings, supra note 9, at 1-2.

<sup>211</sup> See 15 U.S.C. § 1291.

least 25 percent of the new stadium cost and local tax payers paying no more than 25 percent.  $^{212}$ 

According to Specter, SFFRA was a much needed piece of legislation because MLB and NFL teams were demanding publicly financed stadiums from communities "at a time when [these teams] enjoy a [Congressionally created] antitrust exemption [for pooling their sports broadcasts], which has enabled them to have tremendous revenues."<sup>213</sup> Other members of the Judiciary Committee, however, expressed concern about SFFRA. In the opinion of these committee members, SFFRA disparately impacted small market teams that had relied substantially on the league's broadcast revenues, which SFFRA would have required in part diverted to fund new stadiums.<sup>214</sup>

From an outsider's perspective, SFFRA would have been a step in the right direction because it would have prevented the kind of one-sided facility "partnerships" that counties like Miami-Dade ultimately felt compelled to make without such a bill. However, at the same time, the proposed version of SFFRA was insufficient in three respects. First, SFFRA protected communities from MLB and NFL stadium demands, but it did not address arena demands made by NBA and NHL teams. Second, SFFRA allowed communities to continue to pay up to twenty-five percent of the cost of new stadiums — an arbitrary percentage, which is seemingly more than the amount

<sup>&</sup>lt;sup>212</sup> See Weiler & Roberts, supra note 179, at 275; Greenberg, supra note 95, at 394; Marc D. Oram, *The Stadium Financing and Relocation Act of 1999*, 2 VA. J. SPORTS & L. 184, 197-98 (2000).

 $<sup>^{213}</sup>$  Hearings, supra note 9, at 11 (testimony of U.S. Senator Arlen Specter of Pennsylvania).

<sup>&</sup>lt;sup>214</sup> See Hearings, supra note 9, at 57 (testimony of U.S. Senator Russell Feingold of Wisconsin) ("I am particularly concerned with how this legislation would impact small media market teams like the Green Bay Packers."); *id.* at 28 (testimony of U.S. Senator Russell Feingold of Wisconsin) ("[T]he Packers are dependent chiefly on their share of the television revenues from the NFL. In fact, 60 percent of the Packers' operating revenues are derived from these funds. What this means is that this bill's trust fund provision will severely hurt the ability of the Packers to survive.").

<sup>&</sup>lt;sup>215</sup> See supra note 1 and accompanying text.

 $<sup>^{216}</sup>$  See 15 U.S.C.  $\S$  1291 (noting the conspicuous absence of any references about NBA and NHL arenas).

communities would pay if professional sports participated in a truly competitive market.<sup>217</sup> Lastly, SFFRA did not account for the potential situation where a community is not actually subsidizing a sports team but rather operating as a true partner with that team, such as where the community would keep a pro rata share of all revenues derived from the new facility.<sup>218</sup>

# V. DEVELOPING AN APPROPRIATE BILL TO CURB STADIUM SUBSIDIES

Moving forward, it seems the best way for America to protect local communities from sports teams' stadium subsidy demands would be for Congress to pass a bill that removes the Sports Broadcasting Act's limited antitrust exemption from any league in which at least one of its teams accepts facility funding from a local municipality, but does not provide that municipality with a pro rata share of the facility's revenues. A bill of this nature would not outlaw local communities from funding stadium projects; however, it would ensure that no sports team is ever able to "extort" a free ride from its local community.<sup>219</sup>

There are several advantages of this proposed bill. First, this proposed bill would be more effective than relying upon state-based lending of credit and public purpose doctrines because this proposed bill would not lead to a "prisoner's dilemma" with respect to state court enforcement.<sup>220</sup> As history has shown, most state attempts to outlaw stadium subsidies have led to judicially crafted exceptions to keep stadium funding ongoing.<sup>221</sup> The judicial history in Florida is a perfect example of how addressing stadium subsidies on a state level has led to state

<sup>&</sup>lt;sup>217</sup> Edelman, Bargaining Power, supra note 8, at 306.

<sup>&</sup>lt;sup>218</sup> See Stadium Financing and Franchise Relocation Act of 1999, S. 952, 106th Cong. (1999).

<sup>&</sup>lt;sup>219</sup> See Hearings, supra note 9, at 11 (testimony of U.S. Senator Arlen Specter of Pennsylvania).

<sup>&</sup>lt;sup>220</sup> See supra notes 116-53 and accompanying text.

<sup>&</sup>lt;sup>221</sup> *Id*.

courts eventually loosening their own interpretation of "municipal purpose" to avoid losing a sports team.<sup>222</sup>

Second, this proposed bill is a more practical way to address stadium subsidies than court-ordered expansion. Courts in the Third and Ninth Circuits have already rejected ordering leagues to expand to include teams in new communities.<sup>223</sup> In addition, even if any court were to compel expansion, doing so would likely create streams of litigation involving the rights of potential new entrants, as well as logistical concerns about how to schedule games in a league with a constantly changing number of teams.<sup>224</sup>

Third, this proposed bill is fairer than breaking up the four professional sports leagues because it does not force sports teams into accepting a business model that has strong empirical evidence of failing.<sup>225</sup> Although the Department of Justice regularly takes steps to prevent the forming and maintenance of monopolies, there are major distinctions between the professional sports industry and other more traditional including of intra-league industries. the presence competition.<sup>226</sup> These distinctions favor, in the context of professional sports, pursuing market correction through a less drastic remedy.

Finally, this proposed bill is a more effective attempt at solving the real problem of subsidies than is either STADIA or SFFRA. This is because the proposed bill eliminates all

<sup>&</sup>lt;sup>222</sup> See supra notes 135-53 and accompanying text.

 $<sup>^{223}</sup>$  See Seattle Totems Hockey Club, Inc. v. NHL, 783 F.2d 1347 (9th Cir. 1986); Mid-South Grizzlies v. NFL, 550 F. Supp. 558, 560-61, 562 (E.D. Pa. 1982), affd 720 F.2d 772 (3d Cir. 1983).

<sup>&</sup>lt;sup>224</sup> See supra notes 158-78 and accompanying text.

<sup>&</sup>lt;sup>225</sup> See supra notes 179-200 and accompanying text.

<sup>&</sup>lt;sup>226</sup> See Edelman, Single Entity, supra note 3, at 892; see also St. Louis Convention & Visitors Comm'n v. NFL, 154 F.3d 851 (8th Cir. 1998); Sullivan v. NFL, 34 F.3d 1091 (1st Cir. 1994); L.A. Mem'l Coliseum Comm'n v. NFL, 726 F.2d 1381 (9th Cir. 1984); Mid-South Grizzlies, 720 F.2d 772; N. Am. Soccer League v. NFL, 670 F.2d 1249 (2d Cir. 1982); Shaw v. Dallas Cowboys Football Club, Ltd., No. 97 Civ. 5184, 1998 WL 419765 (E.D. Pa. June 23, 1998); McNeil v. NFL, 790 F. Supp. 871 (D. Minn. 1992); but see Am. Needle v. NFL, 538 F.3d 736 (7th Cir. 2008) (finding the NFL is a single-entity in certain licensing markets).

subsidies that do not include a true public partnership.<sup>227</sup> At the same time, the bill allows communities to continue to build sports facilities, as long as they maintain their pro rata share of the facility revenues.

#### CONCLUSION

The current relationship between professional sports teams and American communities needs to change. The unique structure of professional sports leagues, as well as the American government's historic hands-off approach to regulating the sports industry, has allowed America's four premier professional sports leagues to exploit their monopoly power in the market for sports franchises. As a result, professional sports teams have garnered billions of dollars in local subsidies. Governments would otherwise spend these dollars to improve public welfare.<sup>228</sup> Given the excessive power that the sports industry has come to exert over American communities, new efforts must be made to prevent sports teams from continuing to flex their muscles to the detriment of the common citizen.

A simple bill that prevents professional sports teams from accepting a greater share of facility revenues than they pay in construction costs serves as an important step toward curbing professional sports' excessive bargaining power. Similarly, this bill would influence sports leagues to raise capital more efficiently, and would allow communities to focus their tax spending on true public welfare projects.

Professional sports leagues have enjoyed a long-standing positive relationship with the American community. Sports subsidies, however, are a relatively new phenomenon that is detracting from this relationship. Because subsidized professional sports teams hurt public welfare, as well as encourage the inefficient operation of sports teams, a bill preventing sports teams from accepting public subsidies is an

stadium construction, they are moved away from "more important investments like law enforcement, education, and health care.").

<sup>&</sup>lt;sup>227</sup> See supra note 219 and accompanying text.

<sup>&</sup>lt;sup>228</sup> See Hearings, supra note 7, at 111 (testimony of Joseph R. Biden, Jr., United States Senator) (explaining that when tax dollars are funneled toward

important step toward curbing professional sports' bargaining power vis-à-vis the American community.

Placing the professional sports industry back on equal footing with America's local communities would restore an important balance between professional sports and American society. By restoring this balance, American communities would once again become able to enjoy the benefits of professional sports without needing to subsidize the costs of an already profitable industry.